EFFICACY OF ENDOSCOPIC THIRD VENTRICULOSTOMY TO TREAT COGNITIVE SYMPTOMS IN ADULTS WITH CHRONIC OBSTRUCTIVE HYDROCEPHALUS

by

Nicholas Salterio

B.Sc., The University of British Columbia, 2018

A THESIS SUBMITTED IN PARTIAL FULFILLMENT OF THE REQUIREMENTS FOR THE DEGREE OF

MASTER OF SCIENCE

in

THE FACULTY OF GRADUATE AND POSTDOCTORAL STUDIES

(Experimental Medicine)

THE UNIVERSITY OF BRITISH COLUMBIA (Vancouver)

August 2020

© Nicholas Salterio, 2020

The following individuals certify that they have read, and recommend to the Faculty of Graduate and Postdoctoral Studies for acceptance, a thesis/dissertation entitled:

Efficacy Of Endoscopic Third Ventriculostomy To Treat Cognitive Symptoms In Adults With Chronic Obstructive Hydrocephalus

submitted by	Nicholas Salterio	in partial fulfillment of the requirements for		
the degree of	Master of Science			
in	Experimental Medicine			
Examining Co	mmittee:			
Dr. Thomas Zwimpfer, Neurosurgery, Clinical Professor				
Supervisor				
Dr. Dean Foti, Neurology, Clinical Assistant Professor				
Supervisory Committee Member				
Dr. Nicholas Bogod, Neuropsychology, Clinical Assistant Professor				
Supervisory Committee Member				
Dr. Christophe	er Honey, Neurosurgery, Clinical I	Professor		

Additional Examiner

Abstract

Hydrocephalus is a neurological disease caused by an overaccumulation of cerebrospinal fluid (CSF) within the ventricles of the brain. This produces expansion of the ventricles (ventriculomegaly) and compression of the surrounding brain tissue. Obstructive hydrocephalus is the result of a macroscopic blockage within the ventricular system distal to the arachnoid villi and can produce cognitive decline. Endoscopic third ventriculostomy (ETV) has been shown to be a safe surgical treatment in obstructive hydrocephalus, however the efficacy of ETV to improve cognition has limited evidence. In order to determine how ETV affects cognition in adults who have chronic obstructive hydrocephalus, this thesis had the following objectives: 1) to determine the sufficiency of primary ETV to safely improve global cognition in adults with chronic obstructive hydrocephalus, and 2) to determine which cognitive domains ETV produces clinically meaningful changes in adults with chronic obstructive hydrocephalus.

To complete the first objective, we conducted a multicenter study using the Montreal Cognitive Assessment (MoCA) and Symbol Digit Modality Test (SDMT), measures of global cognition and processing speed, before and twice after ETV at 5-months and 14-months follow-up. We found that global cognition was only clinically improved at 14-months follow-up and processing speed was unaffected. Global cognitive worsening was rare at both follow-up assessments.

To complete the second objective, we conducted a single-center study using the RBANS to measure global cognition in better detail and five cognitive domains before and after ETV. We found patients with chronic obstructive hydrocephalus treated with ETV have a clinically and statistically significant improvement in global cognition, attention, and delayed memory domains

iii

at 4-months follow-up. Clinical worsening was rare in all domains except

visuospatial/constructional, where although there was no statistical difference, 50% of patients had a clinical worsening post-ETV. Lastly, a potential predictor of post-ETV global cognitive improvement was discovered. The likelihood of global cognition improvement was 100% when the difference between immediate and delayed memory indexes were three points or less.

Further validation of this predictor is required. Future studies should expand on the neurocognitive testing to include anatomical correlates to elucidate the mechanism of cognitive decline seen in chronic obstructive hydrocephalus.

Lay Summary

Hydrocephalus is a treatable disease that when left untreated, can cause permanent thinkingrelated disabilities and eventually brain swelling, coma, and death. Hydrocephalus can affect all ages. The problem comes from a blockage and backup of the normal fluid spaces that exist within the brain.

An endoscopic third ventriculostomy (ETV), is a brain surgery that can treat the disease and reduce many of the symptoms. There is important information missing about what happens to adults who have an ETV. This is especially true for thinking and memory problems caused by hydrocephalus. We used state-of-the-art thinking and memory tests before and after surgery to figure out that ETVs are safe and improve parts of memory and attention.

Preface

This thesis comprises works with the attempt to determine the efficacy of ETV to improve cognition in adults with chronic obstructive hydrocephalus.

Chapter 2. A version of this material has been submitted for publications as "Zwimpfer, T.J., Salterio, N., Holubkov, R., Katzen, H., Luciano, M.G., Nagel, S.J., Williams, M.A., Wisoff, J.H., McKhann, G.M., Edwards, R., and Hamilton, M.G. *Cognitive and Gait Outcomes after Primary Endoscopic Third Ventriculostomy in Adults with Chronic Obstructive Hydrocephalus*". This study was conducted as a part of the Adult Hydrocephalus Clinical Research Network with Dr. Zwimpfer as the project lead. I was involved in data collection at our center. I designed and managed the analyses and figure design. I wrote the first draft of the manuscript and was involved in editing the final version for submission. Statistical supervision was completed by Dr. Holubkov. All authors reviewed the final version of the manuscript.

Chapter 3. A version of this material is being submitted for publication "Salterio, N., Foti, D., Bogod, N., Katzen, K., and Zwimpfer, T.J. *Chapter 3: Improved Global Cognition, Attention, and Delayed Memory After Endoscopic Third Ventriculostomy in Adult Chronic Obstructive Hydrocephalus Using The Repeatable Battery for the Assessment of Neuropsychological Status (RBANS)*". Dr. Zwimpfer, Dr. Katzen, and I conceived the study. I collected the data, completed the analyses, created the tables and figures, and wrote the first draft of the manuscript. Dr. Foti, Dr. Bogod, and Dr. Zwimpfer provided supervision, assistance with results interpretation, and editorial support. The University of British Columbia Clinical Research Ethics Board approved the study contents of Chapter 2 (H14-02127) and Chapter 3 (H19-01434). The patients provided informed consent for participation in these studies.

Table of Contents

Abstract		. iii
Lay Summary	у	V
Preface		. vi
Table of Cont	ents	viii
List of Tables	•••••••••••••••••••••••••••••••••••••••	Х
List of Figure	S	xi
List of Abbre	viations	. xii
Acknowledge	ments	xiv
Chapter 1: In	troduction	1
1.1 Hydi	rocephalus Epidemiology:	2
1.1.1	The Different Etiologies of Hydrocephalus	3
1.1.2	Symptomatology in Acute and Chronic Obstructive Hydrocephalus	6
1.2 Path	ophysiology of Acute Hydrocephalic Symptomatology	7
1.2.1	Herniation:	7
1.2.2	Respiratory centers:	8
1.2.3	Cardiac centers:	9
1.2.4	Papilledema:	9
1.2.5	Nausea and vomiting:	10
1.2.6	Headache:	11
1.2.7	Loss of consciousness:	12
1.3 Path	ophysiology of Chronic Hydrocephalic Symptomatology	13
1.3.1	Gait dysfunction:	13
1.3.2	Urinary dysfunction:	15
1.3.3	Cognitive dysfunction:	16
1.3.3.1	Definitions of Cognitive Domains	. 17
1.3.3.2	Features of Cognitive Impairment in Chronic Obstructive Hydrocephalus	. 22
1.4 Cere	bral Ventricular Anatomy and Related Neurocognitive Correlates	. 24
1.5 Surg	ical Treatments of Obstructive Hydrocephalus: CSF Shunting vs ETV	. 33
1.6 Test	ing Cognition in Obstructive Hydrocephalus	. 37
1.7 Evid	ence of ETV to Improve Cognition in Obstructive Hydrocephalus	. 38
1.8 Rese	earch Question	41
Chapter 2: Sh	ort- and Long-term Follow-up of Cognitive Outcomes after Primary ETV	in
Adult Chroni	c Obstructive Hydrocephalus	.43
2.1 Meth	nodology	. 43
2.1.1	Inclusion and Exclusion Criteria	.43
2.1.2	Baseline Evaluations	. 44
2.1.3	Intraoperative Data	. 44
2.1.4	Statistical Analysis	. 45
2.2 Resu		45
2.2.1	Patient Population (Figure 1)	43
2.2.2	MOCA (Table 1 and Figure 2)	47
2.2.3	SDM1 (1able 2)	4/
2.3 D1sc	ussion	48

Chapter Third V	r 3: Improved Global Cognition, Attention, and Delayed Memory After /entriculostomy in Adult Chronic Obstructive Hydrocephalus Using Th	Endoscopic e
Repeata	able Battery for the Assessment of Neuropsychological Status (RBANS)	50
3.1	Methodology	50
3.1.	.1 Inclusion and Exclusion Criteria	50
3.1.	.2 Assessments	
3.1.	.3 Intraoperative Data	53
3.1.	.4 Statistical Analysis	
3.2	Results	
3.2.	.1 Patient Population (Figure 3)	
3.2.	.2 MoCA (Table 2 and Figure 4)	55
3.2.	.3 RBANS (Table 2 and Figure 5)	55
3.2.	.4 Pre-ETV Memory Domains as a Predictor of Post-ETV RBANS TS 1 58	Improvement
3.2.	.5 Individual Analysis	
3.3	Discussion	59
3.3.	.1 MoCA	59
3.3.	.2 RBANS TS and TSP	
3.3.	.3 Index Scores	
3.3. scor	.4 Retrospective prediction of global cognition improvement based on r res 66	nemory
3.3.	.5 MoCA vs RBANS	
Chapter	r 4: Conclusions	68
4.1	Summary of Thesis Objectives	
4.2	Objective 1: Conclusions and Limitations	
4.3	Objective 2: Conclusions and Limitations	
Referen	ices	71
Append	lices	97
Apper	ndix A Tables	
Apper	ndix B Figures	100

List of Tables

Chapter 2:
Table 1. MoCA Total Score and Symbol Digit Modalities Test (SDMT)
Chapter 3:
Table 2. Neurocognitive assessments: MoCA and RBANS Domains

List of Figures

Chapter 2:

Figure 1. Flow diagram of patient inclusions, exclusions and follow-up	100
Figure 2. MoCA Total Score for 5-Months and 14-Months Follow-up	101

Chapter 3:

Figure 3. Flow diagram of patient inclusions, exclusions and follow-up	.103
Figure 4. MoCA Total Score for 4-Months Follow-up	.104
Figure 5. RBANS TS, TSP, and Statistically Significant Domains	105
Figure 6. TSP Correlation	.106
Figure 7. Retrospective Global Improvement Prediction Score	.107

List of Abbreviations

AHCRN: Adult Hydrocephalus Clinical Research Network ARAS: Anterior Reticular Activating System **CPC:** Choroid Plexus Cauterization CSF: Cerebrospinal Fluid DLPFC: Dorsolateral prefrontal cortex ETV: Endoscopic Third Ventriculostomy FA: Factional Anisotropy GABA: Gamma Aminobutyric Acid HCRN: Hydrocephalus Clinical Research Network **ICP:** Intracranial Pressure iNPH: idiopathic Normal Pressure Hydrocephalus IQ: Intelligent Quotient MCID: Minimum Clinically Important Difference MMSE: Mini Mental Status Exam MoCA TS: MoCA Total Score MoCA: Montreal Cognitive Assessment MTG: Middle Temporal Gyrus **RBANS TS: RBANS Total Score** RBANS: Repeatable Battery for the Assessment of Neuropsychological Status RCI: Reliable Change Index SDMT: Symbol Digit Modality Test STG: Superior Temporal Gyrus

TPS: Total Percentile Score

Acknowledgements

I would like to thank Dr. Zwimpfer for the more than five years of mentorship and supervision. Your support and guidance have continuously advanced my interest in medicine and clinical research. I appreciate your trust in my roles within your clinical practice and research program. The experiences gained through these years were fundamental to my professional and personal development.

I would like to thank Dr. Foti and Dr. Bogod for sharing their time and extensive knowledge in the fields of neurology and neuropsychology. Their expertise was invaluable as members of my supervisory and examination committee.

I would also like to thank Dr. Christopher Honey and Dr. Serge Makarenko for accepting the role of External Examiner and Chair to providing their extensive research experience to further my education.

I thank all the staff, secretaries, neurosurgery residents, and neurosurgeons for their education over the years.

Lastly, I would like to thank all the patients. I am humbled to have been a part of their care and for their enthusiastic interest and participation in my research projects.

Chapter 1: Introduction

Cognition is the capacity to use prior knowledge to inform current and future decision making. Few causes of cognitive decline have a safe surgical option for treatment and improvement in cognitive symptoms. Hydrocephalus is a category of neurological disease stemming from an overaccumulation of cerebrospinal fluid (CSF) within the ventricles of the brain. This produces expansion of the ventricles (ventriculomegaly) and compression of the surrounding brain tissue (Pong et al., 2017). Obstructive hydrocephalus is the result of a macroscopic blockage within the ventricular system distal to the arachnoid villi and can produce cognitive decline. Endoscopic third ventriculostomy (ETV) has been shown to be a safe surgical treatment in obstructive hydrocephalus (Isaacs et al., 2016; Waqar et al., 2016), yet the efficacy of ETV to improve cognition has limited evidence (Burtscher et al., 2003; Locatelli et al., 2014; Hader et al., 2014; Baroncini et al., 2019).

This chapter will begin with a brief summary of known epidemiology. Description of the different etiologies of hydrocephalus and pathophysiology of symptomatology will follow. Cognitive domains will be defined and a detailed review of ventricular anatomy and potential anatomical correlates to cognition will be described. The surgical treatment options for obstructive hydrocephalus and methods of cognitive testing will be compared. A full review of the current literature of cognition outcomes and ETV in adults will be presented highlighting the gaps that exist. Lastly, the rationale, hypotheses, and objectives of the thesis are stated.

1.1 Hydrocephalus Epidemiology:

An estimated 70,000 Canadian adults currently suffer from an etiology of hydrocephalus (Isaacs et al., 2018). The global prevalence is 84.7/100,000 [95% CI 61.9 to 115.9]. The continents with the highest prevalence include Asia, Africa, and South America. The Asian prevalence is largely due to the increased elderly prevalence of 656.9/100,000 [95% CI 46.6 to 9257.9] compared to Europe and North America's combined, 52.8/100,000 [95% CI 11.8 to 237.0]. The African prevalence of pediatric hydrocephalus is also higher (104.0/100,000 [95% CI 33.3 to 324.77]) compared with the North American (55.6/100,000 [95% CI 41.4 to 74.7]). Prevalence is not uniform across the lifespan with a global pediatric prevalence of 88/100,000, adult prevalence of 11/100,000, elderly prevalence of 175/100,000, and possible prevalence of >400/100,000 above the age of 80.

The incidence of surgical intervention varies based on etiology with idiopathic normal pressure hydrocephalus (iNPH), communicating hydrocephalus, and obstructive hydrocephalus having rates of 2.2 ± 0.8 , 1.9 ± 0.3 , and $0.8 \pm 0.1/100,000/$ year (Sundström et al., 2017). However, the incidence of ETV surgery is only $0.4 \pm 0.1/100,000/$ year compared to 4.7 ± 0.8 shunt surgeries. This shows the disparity of patients with obstructive hydrocephalus that are not receiving ETV surgery.

The initial cost for shunt insertion and ETV is similar, but with shunts requiring more revisions it is more cost-effective to treat obstructive hydrocephalus with an ETV (Reddy et al., 2014; Isaacs et al., 2016; Waqar et al., 2016; Pham et al., 2013). If more obstructive hydrocephalus patients are treated with ETV, this could significantly reduce the estimated \$1.1 billion dollars per year

cost of hydrocephalus treatment in the United States, which a Manitoba Canada study found a similar cost per population (Patwardhan & Nanda, 2005; Del Bigio,1998). As the Canadian population continues to grow, the social and economic burden of hydrocephalus will continue to rise. Effective and cost-efficient treatment modalities, like ETVs, are required to aid in reducing this burden going forward.

1.1.1 The Different Etiologies of Hydrocephalus

Hydrocephalus can be produced by different mechanisms of disease development. There are three main types of hydrocephalus: obstructive/non-communicating, communicating, and iNPH. The major distinction between the types of hydrocephalus is based on where along the ventricular system the obstruction is found. If the obstruction is at the level of the arachnoid villi, where cerebrospinal fluid is absorbed into the venous system, CSF is able to flow through the system unimpeded, but is not adequately absorbed. This is termed communicating hydrocephalus. Conversely, if there is a macroscopic blockage within the ventricular system distal to the villi, then it is termed obstructive or non-communicating hydrocephalus. The pathophysiology for each type of hydrocephalus is varied by the specific etiology and the resulting anatomical changes.

Etiology by level of obstruction:

Obstructive hydrocephalus can be the result of congenital abnormality (aqueductal stenosis or webbing) or malformation (e.g. myelomeningocele, Arnold-Chiari, Dandy-Walker). Aqueductal stenosis or atresia, is a narrowing or absence of the cerebral aqueduct of Sylvius. This reduces or stops the flow of CSF through the aqueduct. Commonly, this is due to developmental

3

malformations and can produce hydrocephalus in utero up-to and including the elderly. The latter termed "arrest hydrocephalus" is a form of unrecognized congenital hydrocephalus as the obstruction has persisted since development, but symptoms only presented in adulthood. The reason for this delay is symptom presentation is currently unproven. It may be due to the inability for neural networks to continue to compensate at a certain degree of age-related atrophy. Tumours (primary benign, neoplasmic and metastases), cysts (arachnoid, colloid), abscesses (rare), within the ventricular system from lateral ventricles blocking foramen of Monro to the outlets of the fourth ventricle can also cause obstructive hydrocephalus through the same mechanism.

Until the recent awareness around folate supplementation during pregnancy, myelomeningocele and its variants were a common pediatric cause of hydrocephalus. Myelomeningocele is an abnormal protrusion of the dural sac with spinal root involvement past the vertebral column. It is often associated with hydrocephalus due to aqueductal stenosis and Chiari malformations (Avagliano et al., 2019). The caudal displacement of the brainstem and commonly malformed tectum (becked tectum) is likely the cause of the narrowing of the aqueduct. Encephalocele is thought to cause hydrocephalus by similar mechanism (Avagliano et al., 2019). Although hydrocephalus was originally part of the Dandy-Walker malformation triad, it is only present in 80% of cases when the cyst causes obstruction within the fourth ventricle (Spennato et al., 2011). Hydrocephalus is usually only seen in syndromic craniosynostosis and is thought to be caused by hypoplastic posterior fossa and skull base venous outlet occlusion (Collmann et al., 2005). Communicating hydrocephalus is the result of microscopic obstruction at the level of the arachnoid granulations, the portion of the neuroaxis that absorbs cerebrospinal fluid into the circulation. Communicating hydrocephalus is generally thought to be an acquired etiology, occurring after an insult to the arachnoid villi. Inter-/intra-cerebral infection (ie. meningitis, ventriculitis, encephalitis) can cause the villi to scar and can result in malabsorption of CSF and results in communicating hydrocephalus. Hemorrhage within the cranial vault can cause hydrocephalus through the same mechanism. Post-traumatic brain injury requiring decompressive craniotomy is associated with hydrocephalus in 11.9-36% who undergo decompressive craniotomy (Fattahian et al., 2018), however the mechanism is still unclear (Satyarthee, 2018)

iNPH is a form of communicating hydrocephalus but is not associated with an increase in intracranial pressure (ICP). This form of hydrocephalus is chronic and occurs by definition over the age of 60 (although debate exists on the appropriate age for classification) without meeting the criteria for any of the aforementioned etiologies (Marmarou et al., 2005; Mori et al., 2012; Halperin et al., 2015; unpublished AHCRN guidelines). Although the mechanism behind it is unknown, there have been some hypotheses presented based on vascular, hemodynamic, and metabolic aspects (Bräutigam et al., 2019). iNPH presents chronically with a well-defined clinical finding known as Hakim's triad: gait, urinary, and cognitive dysfunction.

In rare instances, choroid plexus hyperplasia (23 cases reported) and choroid plexus papilloma/carcinoma have been suggested to cause an overproduction and lower absorption of CSF leading to hydrocephalus (Hallaert et al., 2012).

For completeness, there exists a third population of patients who appear to have both communicating and obstructive forms, but there is no method of determining if a patient has both forms until after surgical treatment. This is likely the reason why approximately 13-27% of ETV fail (Isaacs et al., 2016; Waqar et al., 2016). The obstruction to CSF flow has been bypassed with a patent ETV, however there is reduced absorption at the arachnoid villi requiring shunting of the CSF to a different body cavity.

1.1.2 Symptomatology in Acute and Chronic Obstructive Hydrocephalus

Obstructive hydrocephalus patients can present acutely with raised ICP symptomatology. Alternatively, chronic obstructive hydrocephalus can present with normal pressure symptoms (Hakim's triad) alone or in combination with some symptoms from raised ICP. It is important to note that although chronic obstructive hydrocephalus may present similarly to iNPH and results in chronic ventriculomegaly, the etiology of the disease and preferred treatment is different. Thus, the presentation of symptoms severity and efficacy of treatment cannot be equivocated to those of iNPH. iNPH has an idiopathic cause of CSF malabsorption, whereas chronic obstructive hydrocephalus has a defined macroscopic obstruction limiting the amount of CSF that is able to be absorbed. It likely for this reason why ETV has been shown to be ineffective in iNPH in a randomized controlled trial and systematic review (Tudor et al., 2015).

1.2 Pathophysiology of Acute Hydrocephalic Symptomatology

The symptomatology of hydrocephalus is varied by the acuity of symptom onset, with the most acute presentations consisting primarily of raised ICP symptoms. As the presentation becomes more chronic, normal pressure symptoms dominate.

Raised intracranial pressure symptoms vary in severity from headache to nausea and vomiting, loss of consciousness, papilledema, brain herniation, coma, and eventually death. The mechanism for these symptoms is based off the Monro-Kellie hypothesis, which states that there is a defined and limited volume within the skull (Monro, 1783). The brain parenchyma, blood, and CSF each make up a part of this volume. If any of the three increases in size (tumour, hematoma, CSF backup in hydrocephalus) the total volume will exceed the volume available within the skull. The compensatory mechanism is to raise ICP to force more fluid (blood, CSF, extracellular fluid, and intracellular fluid) out of the skull. Once this compensation has been exhausted and ICP continues to increase, pressure gradients between different compartments of the brain within the skull emerge. Namely, the falx cerebri and tentorium cerebelli. With raised ICP, herniation of different components occurs.

1.2.1 Herniation:

The three types of herniation are as follows: transtentorial, subfalcine, and tonsillar (Dunn, 2002). Transtentorial herniation can be subdivided into uncal or central. Uncal herniation occurs when the uncus of the temporal lobe, the medial extent of the parahippocampal gyrus, is pushed under the tentorium and places pressure onto the midbrain of the brainstem.

Central herniation consists of diencephalon (thalamus, hypothalamus, subthalamus, and epithalamus) being pushed under the central (medial) aspect of the tentorium. Again, pressure on the midbrain results.

Subfalcine or cingulate/transfalcine herniation comprises the frontal aspect of the cingulate cortex being pushed laterally under the falx cerebri into the contralateral hemisphere. Unlike the other forms of herniation, subfacline does not result in brainstem compression. Instead, anterior cerebral artery compression can occur leading to an ischemic injury to the medial and paramedial aspects of the hemispheres including the frontal poles. Subfacline herniation can also result in central herniation.

Tonsillar herniation is the result of the tonsils of the cerebellum being pushed through the foramen magnum. Compression of the lower medulla and upper cervical cord can lead to dysfunction of the cardiac and respiratory centers within the medulla. These forms of herniation account for the later phases of raised ICP, coma, and death.

1.2.2 Respiratory centers:

The respiratory centers of the brain are controlled by the three groups of nuclei: the dorsal, ventral, and pontine respiratory groups. The dorsal and ventral groups are located within the medulla, and the pontine group is found within the pons (Saether et al., 1987). The dorsal group is responsible for inhalation and under the influence of the solitary nucleus. The ventral group is responsible for exhalation and is under the influence of both the nucleus ambiguus and retro-ambiguus nucleus. The pontine group is divided into the pneumotatic and the apneustic center,

which are interconnected and are under another component of the solitary nucleus. The pneumotaxic center manages breathing rate and pattern of the respiratory system (Song et al., 2006). The apneustic center manages the intensity of breathing. Therefore, dysfunction of this system leads to depression and eventual cessation of breathing, causing a hypoxic environment throughout the brain and body. When this occurs, intubation and forced oxygen via machine ventilation is mandatory for sustained life.

1.2.3 Cardiac centers:

The cardiac centers of the brain are contained within the solitary nucleus, ventrolateral depressor area, and the ventrolateral pressor area (Maeda et al., 1994). The solitary nucleus is located in the dorsal medulla, while as their names suggest, the ventrolateral depressor and pressor areas are found more ventrally within the medulla. This is slightly counterintuitive as ventral medulla is typically thought of as purely motor fibers running down the pyramids. However, just posterior to the pyramids in the rostral medulla are where these ventrolateral areas are found. Originally, the cardiac centers were thought to only control the regulation of systemic blood pressure and heart rate. But, Maeda et al. (1994) showed that these centers also control cerebrovascular vasoconstriction. This could explain one of the causes of brain ischemia with raised ICP.

1.2.4 Papilledema:

Papilledema is the swelling of the optic disk. The optic nerves can be thought of as extensions of the brain. Papilledema in adults is 100% sensitive and 98% specific for raised ICP (Tuite et al., 1996). The optic canal is the bony structure where the optic nerve and ophthalmic artery reside, running along the skull base before entering the orbital cavity. They are surrounded by the optic

sheath made of dural and arachnoid layers. The subarachnoid space and the optic sheath are communicating structures with respect the CSF flow (Hayreh, 1984). As ICP increases, the axonal fibers within the sheath become damaged and a reduced axoplasmic flow of the optic disk results in papilledema (Schirmer & Hedges, 2007). This can cause transient reduced visual acuity and chronically can cause permanent blindness if left untreated.

1.2.5 Nausea and vomiting:

The pathophysiology of nausea and vomiting associated with raised ICP is related to pressure being put onto the medullary structure responsible for emesis, the area postrema (Irwin & Rippe, 2008; Maule, 1990). This can occur with tonsillar herniation as mentioned above. The area postrema is located just superior and ventral to the obex of the medulla (Miller & Leslie, 1994). The obex can be thought of as the transition zone between the floor of the fourth ventricle and the ventral aspect of the central canal. This cervicomedullary junction is at the level of the foramen magnum and thus can be compressed with tonsillar herniation.

The area postrema's primary connection is to the solitary nucleus. This nucleus has many connections to other autonomic centers within the brainstem. Together the area postrema and the solitary tract are the primary centers for autonomic emesis control. When functioning normally, both these centers have permeable capillaries, which allows for gross autonomic regulation of hormones, proteins, and toxins. Specifically, within the area postrema, the chemoreceptor trigger zone is located. It is this area that controls nausea and vomiting activation. In response to an emetic stimulus, the chemoreceptor trigger zone is activated which has outputs to the nucleus of the solitary tract. From here, more than 25 different nuclei may be activated to coordinate the

10

complex motor movement of vomiting. This is the start of the final common pathway to initiate vomiting from any emetic stimulus, be it an emetic drug or compression from raised ICP. Two known and important components of this pathway are the activation of the retrofacial nucleus and the dorsal motor nucleus of the vagus cranial nerve. The former is the impetus for the required respiratory pattern of emesis. The latter connects to the gastrointestinal system and prepares it for emesis. Additionally, secondary neuronal activation of the trigeminal, facial, and glossopharyngeal cranial nerves occurs for the motor action of emesis.

1.2.6 Headache:

Headache associated with raised intracranial pressure is thought to be due to the pain receptors on the meninges (including the falx) surrounding the brain (Yamada et al., 2018). As the brain itself does not contain pain receptors, the expansion of the brain parenchyma against the meninges and the inner plate of the skull causes the meninges to stretch. This stretching is thought to be an irritant to the meningeal nerves and the impetus of the headaches experienced with raised ICP. A good example is the "thunderclap" headache associated with a subarachnoid hemorrhage. A large amount of bleeding into the cranial vault causes quick expansion of the arachnoid and dura. The nociceptors along the tentorial nerve, the first branch of the ophthalmic branch of the trigeminal nerve that innervate the meninges, are activated by the stretching. This signal is relayed to the subnucleus caudalis of the spinal trigeminal nucleus, where pain is processed. From here, the signal travels along the anterior trigeminal-thalamic tract to the contralateral ventral posterior medial nucleus of the thalamus and continues on to the primary somatosensory cortex. This cortex has been shown to be involved in both pain perception and modulation, given this pain circuit (Bushnell et al., 1999).

11

1.2.7 Loss of consciousness:

Loss of consciousness in an acutely raised ICP scenario is a result of brain herniation onto the brainstem's nuclei responsible for maintaining consciousness. This grouping of more than 20 nuclei is referred to as the anterior reticular activating system (ARAS) and is located within the superior tegmentum (midbrain and superior pons) of the brainstem. Multimodal sensory input from the spinoreticular pathway as well as the olfactory, optic, trigeminal, and vestibulocochlear nerves synapse onto the various nuclei within the ARAS (Iwańczuk & Guźniczak, 2015). These sensory inputs all reach the diencephalon but can travel directly through the medial column of the reticular formation or diffusely through non-specific pathways. Within the diencephalon, these projections separate into paths towards the thalamus and hypothalamus. At the level of the thalamus, the reticular and interlaminar thalamic nuclei act as the relay station between the brainstem and cerebral hemispheres. The reticular thalamic nuclei are mostly specific to pain signals and the interlaminar thalamic nuclei projects the ARAS signals to the cortex, forming part of the thalamocortical network. This network is one of the primary excitatory signals to the cerebral cortex, which at its baseline state is heavily depressed via gamma aminobutyric acid (GABA) inhibition. Therefore, with compression to the ARAS from brain herniation, excitatory signals are reduced to the cerebral cortex leading to a lower level of consciousness. If severe, longstanding, or bilateral this leads to coma and eventually death. Death results from an unprotected airway, requiring intubation to force oxygen into the lungs.

Additionally, raised ICP can lead to cerebral hemisphere ischemia due to decreased cerebral perfusion pressure in response to raised ICP (Bor-Seng-Shu et al., 2013). If both hemispheres

become dysfunctional then consciousness can be impaired. This is due to widespread apoptosis and autophagy of neuronal and vascular cells within the cerebral hemispheres. Many circuits and networks will become dysfunctional, leading to eventual brain death.

1.3 Pathophysiology of Chronic Hydrocephalic Symptomatology

As the acuity of the initial presentation of hydrocephalic symptoms lessens, normal pressure symptoms dominate. These chronic symptoms can be easily distilled into the classic triad: gait, urinary, and cognitive dysfunction.

1.3.1 Gait dysfunction:

The prevailing theories surrounding gait dysfunction in iNPH are centered around the subcortical and periventricular white matter, deep grey nuclei, intracortical circuits, and midbrain's mesencephalic locomotor region (MLR), and recent work on circulation and neurotransmitter involvement.

The corticospinal tract runs along the edges of the lateral ventricles. These fibers control leg movement (Lee et al, 2005; Mocco et al, 2006). The corticospinal tract starts with activation of Betz cells in layer V of the primary motor cortex. The axons travel through the corona radiata, around the lateral ventricles to the posterior limb of the internal capsule. They continue through the crus cerebri, to the pyramids and decussate before entering the superior cervical spinal cord. The lumbar and a portion of the sacral spinal cord lead to efferent motor nerves that stimulate the musculature in the legs, causing movement. Due to the proximity of the corticospinal tract within the corona radiata to the lateral ventricles, compression or atrophy of these upper motor neurons could result in gait dysfunction. MRI analyses of corticospinal tract have found evidence of axonal compression, stretching, and damage in the corticospinal tract which is partially reversed post-shunting (Assaf et al., 2006; Hattingen et al., 2010, Jang et al., 2011, Hattori et al., 2011, Scheel et al., 2012; Jurcoane et al., 2014; Kamiya et al., 2017; Irie et al., 2017; Ades-Aron et al., 2018)

Intracortical connection dysfunction in the hydrocephalic patient is a new line of evidence for the mechanism of gait dysfunction. Chistyakov et al. (2012) used transcranial magnetic stimulation over the medial aspect of the primary motor cortex (ie. lower extremity representation) to show that resting motor threshold was increased and short intracortical inhibition was decreased prior to surgical treatment in iNPH patients. This resulted in an overexcited primary motor cortex, leading to disturbances in corticospinal motor outputs. After ventricular shunting, both these metrics improved along with expected gait improvement. This study provided initial evidence that upper motor neuron dysfunction via intracortical connection compression from lateral ventricle ventriculomegaly may play a role in the gait dysfunction seen in iNPH patients.

The MLR is located proximal to the cuneiform nucleus, and ventral to the inferior colliculus. It is suggested that the pontine tegmental nucleus and extrapyramidal area are the functional areas within the MLR. Grossly, the MLR can be divided into a dorsal and ventral portion. The dorsal extent is paramedial and ventral to the periaqueductal grey encasing the cerebral aqueduct of Sylvius. The ventral extent is further ventral and lateral. Sherman et al. (2015) showed that the dorsal portion of the MLR is responsible for locomotion and the ventral portion for standing. Due to the closeness of these nuclei to the aqueduct, it is logical to suggest that enlargement of

14

the aqueduct could cause dysfunction of these nuclei and lead to gait dysfunction. Imaging studies using MRI, CT, and newer modeling techniques show conflicting evidence of atrophy in these areas (Lee et al., 2005; Mocco et al., 2006; Hiraoka et al., 2011). Cadaver based studies should be performed to provide insight into the precise anatomical changes in a hydrocephalic patient. Additionally, patients with obstructive hydrocephalus with obstruction rostral to the anterior aspect of the cerebral aqueduct should not display this compression. Follow-up studies comparing these populations should also be conducted.

A handful of hypotheses have been suggested to explain the gait dysfunction seen in the hydrocephalic patient, from simple corticospinal tract dysfunction as ventricular size increases, to intracortical and midbrain nuclei compression. However, a causal mechanism is yet to be determined.

1.3.2 Urinary dysfunction:

Unfortunately, urinary dysfunction is understudied in hydrocephalus. It is a common notion that urinary dysfunction is a side effect of aging. However, urinary dysfunction is always pathogenic and can typically be improved with treatment of the underlying cause (Rigamonti, 2014). The study methods are also a deterrent for researchers and participants; urinary logs to the weighing of soiled pads. As such, the pathophysiology behind urinary dysfunction is limited in chronic hydrocephalus. The current hypothesis is non-specific to hydrocephalus as it is directly taken from other diseases. Micturition (or urinary) reflexes are the set of neural loops that control urination. They are dependent on the spino-bulbo-spinal reflex (Lee et al., 2011). Suprapontine structures including the frontal cortex and basal ganglia suppress the pontine centers via the periaqueductal grey matter that surrounds the cerebral aqueduct of Sylvius. This descending information synapses onto the pontine micturition center (Barrington's nucleus) within the rostral pons. When it is activated, motor neurons through the spinal cord trigger concomitantly the smooth muscle constrictors of the bladder and the striated muscle dilators of the urinary sphincter allowing for micturition. Moreover, sensations of a filled bladder ascends the spinal cord to the pontine mictures, completing the reflex loop.

Since the suprapontine networks are not well understood, there is the potential for the ventriculomegaly of the lateral and third ventricles to disrupt part of this network and cause downstream effects. Alternatively, cerebral aqueduct enlargement may disturb the periaqueductal grey matter, which acts as a rely between ascending and descending urinary information. This would account for why patients with hydrocephalus display both incontinence from being unaware of the filled bladder and frequency symptoms. However, at this time, no evidence has been put forth supporting these ideas. More work is needed to understand this aspect of Hakim's triad.

1.3.3 Cognitive dysfunction:

Cognitive dysfunction has no specific mechanism identified for symptom development in hydrocephalus. First, definitions of cognitive domains will be presented. Then, a description of

the features of cognitive impairment found in chronic obstructive hydrocephalus will be reviewed from a neuropsychological perspective. The following section will review potential anatomical correlates to these cognitive domains in the context of ventriculomegaly. Together they will form a framework from which to study cognitive change in chronic obstructive hydrocephalus post-ETV.

1.3.3.1 Definitions of Cognitive Domains

To assess the current neuropsychology literature surrounding chronic obstructive hydrocephalus and ETV, a clear understanding of the domains tested is required. There are six cognitive domains commonly tested in this literature: executive function, attention, immediate memory, delayed memory, visuospatial/constructional, and language.

Executive Functions

Executive functions combine a collection of complex processes in a system that provides supervision atop the brain's hierarchical processing structure, especially in novel contexts (Strauss et al., 2006). In addition, executive functions contain the skills required for purposeful, goal-directed behaviour. This allows individuals to create strategies specific to the context and evaluate their effectiveness. Executive dysfunction may consist of inappropriate social behaviour, poor decision making and judgment, and planning and organizational difficulties. Executive, attentional, and immediate memory dysfunction may coexist together causing executive function problems in a memory context (ie. memory of prior planning), distractibility, or poor task shifting.

Attention

Attention is the intrinsic quantity and speed limiter for processing information in the brain (Strauss et al., 2006). It can be thought of as a gateway for information flow (Cohen, 1993). Conceptualization of attention has resulted in many different models of attention being proposed (Mesulam, 1981; Posner & Petersen, 1990; and Mirsky, 1996). Mirsky's (1996) model of attention was formed from validated commercially available neuropsychological assessments and is the only model to have putative neuroanatomy from lesional studies for each element. The same elements have been found through factor-analytic studies (Kremen et al., 1992; Mirsky et al., 1991; Strauss et al., 2000). Mirsky's elements of attention include focus-execute (selective attention and rapid perceptual-motor output), shifting (change attentional focus, cognitive flexibility and adaptability), sustained/stable (vigilance and consistency of attentional effort), and encode (short-term memory; Kent, 2016) (Zillmer et al., 2008).

Additionally, arousal can be linked to attention as the prerequisite for attentional processing (Zillmer et al., 2008). Detailed anatomy pertaining to arousal and loss of consciousness can be found in *Section 1.2.7 Loss of Consciousness*.

Memory

Memory is the process of encoding (storing), consolidation (strengthening), and retrieving information (Strauss, 2006). Memory is often modelled based on specific subdivisions relating to the type of information to be remembered. The first subdivision of memory is into short-term and long-term memory.

Short-term memory, immediate memory, working memory and attentional encoding are often conflated (Kent, 2016). Short-term memory is specific to encoding information before consolidation and is made up of sensory memory (registration of sensory inputs) and immediate memory (initial phase of temporary storage of sensory memory) (Lezak et al., 2012). Immediate memory last between 30 seconds to several minutes. Short-term memory can also last for up to two days through different mechanisms. Working memory is specific to processing information and can involve short-term and long-term memory components, as such it is associated with executive functioning. The encoding element of attention is temporary storage of information through a "span of attention" and is therefore short-term memory. Sensory inputs difference (Strauss et al., 2006) and theoretical models of working memory have also been proposed (Baddeley & Hitch, 1974), but are unnecessary for the contents of this thesis.

Long-term memory or delayed memory contains permanent or extended storage of information. It is first divided into declarative/explicit memory and non-declarative/implicit memory. Explicit memory contains information that requires conscious awareness to remember (Zillmer et al., 2008; Camina & Güell, 2017). This can be further divided into episodic and semantic memory (Zillmer et al., 2008). Episode memory contains autobiographical information with spatial and temporal components. Semantic memory contains factual based information without temporal components.

Implicit memory contains certain actions and skills, and unconscious memories (Camina & Güell, 2017). It is better defined through each of the four types of implicit memory: procedural, associative, non-associative, and priming. Procedural memory contains the motor and executive

function information required to perform a specific skill. Associative memory is simply the result of classical and operant conditioning. Non-associative memory is evolutionarily one of the more basic forms of memory resulting from habituation and sensitization. Habituation is when repeated stimulus leads to a decrease in response. Sensitization is the opposite, where repeated stimulus leads to an increase in response. Priming occurs when a primary stimulus alters the future response to a secondary stimulus after the primary stimulus has been forgotten from conscious recall.

Visuospatial and Constructional

Visuospatial and constructional condition is defined as the ability to break down an object or picture into parts (visuospatial perception) and construct a replica based on these parts (constructional) (Mervis et al., 1999). It also requires integration from the attention and executive functions domains (Biesbroek et al., 2014).

Language

The classical model of the language domain consists of Broca's area for speech production, Wernicke's area for language comprehension and the arcuate fasciculus as the white matter connection (Fujii et al., 2016). However, as our understanding of language has increased the classical model has required expansion. The current accepted model of language consists of the dorsal and ventral streams of language processing. The dorsal stream is associated with phonological processing. The ventral stream is associated with semantic processing. The anatomy associated with the parts of language will not be discussed beyond this section, therefore a review of the dorsal and ventral systems is found below.

20

Language processing is initiated with low-level auditory processing at Heschl's gyrus (Middlebrooks et al., 2016; Fujii et al., 2016). Anterior, middle, and posterior (Wernicke's area) aspects of the superior temporal gyrus (STG) perceive phonemes from speech sounds from Heschl's gyrus and is the beginning of the dorsal stream. The middle STG has been shown to contain a phonetic alphabet. Therefore, bilateral damage to this area results in pure word deafness. After phoneme perception, phonological retrieval occurs in the posterior STG and posterior superior temporal sulcus. Since these are large areas of Wernicke's area, Wernicke's aphasia results from damage in the dominant hemisphere. Next, the supramarginal gyrus is responsible for preservation of phonologic information and verbal working memory. The ventral segments of the arcuate fasciculus connects the STG and supramarginal gyrus along the superior longitudinal fasciculus to the frontoparietal operculum including the pars opercularis (Broca's area) and ventral premotor cortex, and in 40% of people to the pars triangularis. The pars opercularis is responsible for syntax. The ventral premotor cortex is important for speech output and phonological processing bilateral and results in dysarthria/anarthria when damaged in 83% of left and 55% of right hemispheres in patients (Tate et al., 2014). Pars triangularis (Broca's area) functions as both semantic and phonologic processing center along an anterior-posterior orientation.

The ventral stream initiates in the middle temporal gyrus (MTG) (Middlebrooks et al., 2016; Fujii et al., 2016). The posterior MGT is responsible for lexical-semantic processing (bilaterally) and prosody (non-dominant hemisphere) (Karunanayaka et al., 2007; Friederici et al., 2015). Moving anteriorly along the MGT is associated with more semantic processing. The inferior temporal gyrus is also thought to be involved in semantic processing. The temporal pole is involved in the semantics of living things (proper names and famous face naming), theory of mind, and sentence-level semantics (Binder et al., 2009; Bi et al., 2011; Binder et al., 2011, Barnett et al., 2014). The angular gyrus is responsible for syllable discrimination, identification, and lexical-semantic processing (Yagmurlu et al., 2016). The inferior fronto-occipital fasciculus connects the MTG, interior temporal gyrus, and angular gyrus to the dorsolateral prefrontal cortex (DLPFC), pars orbitalis, and pars triangularis. The DLPFC is an integration center and is responsible for the semantic stream of language and verbal working memory (Binder et al., 2009; Yagmurlu et al., 2016). The pars orbitalis is connected to the semantic network however the exact function has not been elucidated.

Dorsal and ventral streams can be examined through neurocognitive testing of picture naming and semantic fluency, respectively. Additionally, orthography follows a different stream and can be individually tested through different assessment but will not be discussed in this thesis (see Middlebrooks et al., 2016 for a detailed anatomical review).

1.3.3.2 Features of Cognitive Impairment in Chronic Obstructive Hydrocephalus

Cognitive dysfunction in chronic hydrocephalus has been shown to follow a frontal-subcortical pattern (Ogino et al., 2006). Chronic obstructive hydrocephalus lacks large studies to provide the specific cognitive domain deficits. Of the studies that use multi-domain cognitive testing, 70-100% patients had a deficit in at least one of the following cognitive domains: executive, attention, immediate memory, delayed memory, visuospatial/constructional, and language either
pre- and/or post-ETV (Burtscher et al., 2003; Lacy et al., 2009; Al-Jumaily et al., 2012; Hader et al., 2014).

Burtscher et al. (2003) found that none of the six patients with unrecognised congenital obstructive hydrocephalus had fully intact cognition pre-ETV. Overall, two patients had isolated memory impairments with the remainder having multiple impaired domains. Episodic verbal learning and memory was impaired in five patients, recognition was impaired in two patients, figural memory was impaired in four patients, executive function (including verbal fluency, cognitive flexibility, or attention) was impaired in three patients, and visuospatial/constructional impaired was found in one patient. Orientation and auditory working memory were preserved in all patients. Locatelli et al. (2014) found 8/13 patients with unrecognized congenital aqueductal stenosis had a pre-ETV cognitive deficit. Hader et al. (2014) found 11/13 patients with chronic obstructive hydrocephalus had at least one cognitive domain with borderline (≤ 16 percentile) or impaired (≤ 7 percentile) performance. They also found a mean percentile of 51 for intelligence with no patient having borderline or impaired performance. Another study used the MMSE in 15 patients with chronic obstructive hydrocephalus and found a pre-ETV mean of 25.7/30 (Baroncini et al., 2019). However, the MMSE had been shown to be an insensitive measure of the pattern of cognitive deficits observed in hydrocephalus (Golomb et al., 1994; Iddon et al., 1999).

1.4 Cerebral Ventricular Anatomy and Related Neurocognitive Correlates

Knowledge of the cerebral ventricles, CSF, and meninges has been known for millennia. The latter two were described in an Egyptian papyrus dated approximately 3500 BC (Schiller, 1997; Clarke & O'Malley, 1996).

"Smashing his skull...opening to the interior...the membrane enveloping his brain so that it breaks open his fluid in the interior of his head."

Aristotle (384-322 BC) mentions the "hollow inside as well as of some membrane surrounding it". Herophilus (290 BC) and Erasistratus (280 BC) are credited with the first full descriptions of the four ventricles and their connections to each other.

Contemporary gross description of the ventricular system is based off four fluid filled cavities within the brain (Standring, 2008). The two lateral ventricles (previously termed first and second ventricle) are each found within a cerebral hemisphere. The foramen of Monro/interventricular foramen connects the lateral ventricles to the third ventricle. The aqueduct of Silvius connects the third ventricles to the fourth ventricle located posterior to the brainstem. The fourth ventricle has three apertures and is continuous inferiorly with the central canal of the spinal cord.

The lateral ventricles can be further divided into five sections from anterior to posterior: frontal/anterior horn, body, trigone/atrium, temporal/inferior horn, occipital/posterior horn.

Frontal Horns of Lateral Ventricles

The frontal horns lie within the frontal lobe of the cerebral hemisphere (Standring, 2008). The anterior wall and roof are formed by the genu and anterior body of the corpus collosum. The frontal horns are separated medially by the septum pellucidum. The posterior boundary is the foramen of Monro where the continuous body of the lateral ventricles commences. The floor mainly consists of the head of the caudate nucleus with the rostrum of the corpus callosum forming a section of the medial aspect. Deep to the caudate nucleus is the internal capsule and lenticular nucleus.

The dorsolateral prefrontal circuit emerges from the dorsolateral prefrontal cortex (Broadmann's area 9 & 10) and projects to the dorsolateral head of the caudate nucleus (Tekin & Cummings, 2002). These fibers continue to the lateral dorsomedial globus pallidus interna and rostrolateral substantia nigra pars reticulata. Pallidal fibers project to the ventral anterior and mediodorsal thalamus, which returns fibers to the dorsolateral prefrontal cortex. Disruptions anywhere in this circuit results in dorsolateral prefrontal syndrome resulting in executive cognitive dysfunction. This can result in poor organizational strategies, poor memory search strategies, environmental dependency, impaired set shifting, and verbal/manual dissociation.

Specific disruption along the circuit results in certain cognitive ability dysfunction (Tekin & Cummings, 2002). Disruption along the DLPFC and caudate results in poor recall with preserved recognition. Pallidal disruption, like in Huntington's and Parkinson's disease, also results in retrieval type memory loss. The thalamus is involved in both the temporal-limbic and frontal-subcortical circuits, so disruption results in poor recall and poor recognition causing an amnestic syndrome. As previously mentioned, the head of the caudate forms the floor of the frontal horns

25

and is therefore susceptible to compression which may resulting in the specific cognitive dysfunction of poor recall with preserved recognition.

In adults with iNPH, ventriculomegaly resulted in displacement of caudate nuclei and decreased volumes associated with worse neurocognitive performance and severity of neuropsychiatric symptoms (Peterson et al., 2019). Abnormal regional network parameters were also significantly different in pediatric hydrocephalus patients in many regions including the medial frontal gyrus for all parameters and caudate nucleus for a subset of parameters (Yuan et al., 2015). Chronic hydrocephalic patients were also found to have an increased fractional anisotropy (FA) value, indicative of compression, in the caudate nucleus compared to controls with ex vacuo ventriculomegaly (Osuka et al., 2010). Moreover, FA values returned to normal post-shunting. An experimental study showed that kaolin induced hydrocephalus in rats caused decreased cortical gray matter thickness with microstructure rearrangement, decreased caudate-putamen cross-sectional area, compression of caudate-putamen and ventral internal capsule, and alternation of corpus callosum and periventricular white matter microstructure (Lugé et al., 2016). These three studies provide evidence of disruption in regions of the dorsolateral prefrontal circuit and associated changes in neurocognitive testing. This is suggestive of a potential neuroanatomical correlate for some of the cognitive dysfunction seen in hydrocephalic patients.

Bodies & Atria of Lateral Ventricles

The bodies extend throughout the frontal and parietal lobes (Standring, 2008). The septum pellucidum continues to separate the lateral ventricles throughout the body. The body of the fornix forms the inferior limit and medial wall of the body. Lateral to the body of the fornix is

26

the thalamus, which are separated from each other by the choroidal fissure containing the choroid plexus and superior choroidal vein. The choroid plexus produces CSF. ETV and choroid plexus cauterization (CPC) has been shown to be effective in African pediatric post-infectious hydrocephalus, but the Hydrocephalus Clinical Research Network (HCRN) has found higher revision rates compared to ETV alone (Kulkarni et al., 2017; Pindrik et al., 2020).

The lateral walls of the bodies of the lateral ventricles are formed from the thalamus inferiorly and body of the caudate nucleus superiorly (Standring, 2008). The groove between the caudate nucleus and thalamus contains the stria terminalis and thalamostriate vein. The stria terminalis is a white-matter bundle and major output pathway of the amygdala to the septal nuclei, ventromedial nucleus of the hypothalamus, and thalamic areas. It is thought to serve an important role in delayed or unpredictable threats (~10 minute) stress responses including cardiovascular responses and a potential site of deep brain stimulation for severe obsessive-compulsive disorder (Goode et al., 2020; Gomes-de-Souza et al., 2020; Luyten et al., 2016). The thalamostriate vein is an important landmark for intra-operative ventricular navigation as it marks the direction of the lateral wall of the body from the foramen of Monro. This allows for identification of which lateral ventricle the endoscope has entered.

As the bodies widen from anterior to posterior and the continuous temporal and occipital horns start to form, the central aspect connecting these three areas together is the atrium (Standring, 2008).

Occipital Horns of Lateral Ventricles

The occipital horns extend past the atrium into the occipital lobe (Standring, 2008). The splenium of the corpus callosum (forceps major) forms the upper medial wall, however the calcar avis (previously hippocampus minor) forms the majority of the medial wall which separates the occipital horn from the calcarine fissure (Flores, 2002). The most common floor of the inferior horn consists of the collateral eminence, the white matter associated with the indentation of the collateral sulcus (97% of people), but variations including the inferolateral surface from the occipitotemporal sulcus, both collateral eminence and inferolateral surface and neither can exist as well (Switka et al., 1999). The tapetum of the corpus callosum (most posterior aspect of forceps major) forms the roof and lateral wall of the occipital horns. The tapetum connects the hippocampi bilaterally (Maller et al., 2019). Limited research has shown the tapetum in patients with Alzheimer's disease to have reduced FA and increased mean diffusivity in diffusion weighted imaging indicative of neurodegeneration (Mayo et al., 2016). FA was also reduced in the tapetum of posttraumatic stress disorder patients, as hippocampal dysfunction is consistently associated with this disorder (Dennis et al., 2019). This limited evidence may point to another potential correlate to hippocampal memory dysfunction as found in hydrocephalus and other diseases.

Temporal Horns of Lateral Ventricles

Lastly, the temporal horn extends infero-laterally from the atrium into the temporal lobe around the pulvinar nucleus of the thalamus (Standring, 2008). It extends anteriorly to within 2.5 cm of the temporal pole adjacent to the uncus. The superior temporal sulcus can be used as a cortical landmark for the temporal horn. The roof from medial to lateral consists of the stria terminalis, tail of caudate nucleus, and tapetum which extends laterally to form the lateral wall. The floor is formed by the collateral eminence. The medial wall is formed by the alveus of the hippocampus with CA 1 deep to it. The alveus is a thin layer of white matter with ependymal lining that form the ventricular surface of the hippocampus, it is continuous with the fimbria hippocampi. Hippocampus volume has been associated with verbal learning abilities (Aslaksen et al., 2018; Bähner et al., 2017 & Antoniades et al., 2017; Ajilore et al., 2015; Cao et al., 2016). Experimental hydrocephalus animal models have shown volumetric, neurotransmitter, and morphological changes in the hippocampus suggestive of compression and axonal apoptosis (Kondziella et al., 2009; Taveira et al., 2012). Expansion of temporal horn and hippocampal dysfunction may underlie short-term memory problems seen in hydrocephalic patients.

Foramen of Monro

This crescent or oval passageway between the lateral ventricles and the third ventricle was discovered by Alexander Monro (1733-1817), who is also known for creating the Monro-Kellie doctrine (Tubbs et al., 2014). The superior border consists of the body of the fornix posteriorly and one of the columns of the fornix traveling in an arc in an anterior-inferior fashion. The posterior-inferior aspect is formed from the convex of the anterior pole of the thalamus (Tubbs et al., 2014). The choroid plexus and branches of the medial posterior choroidal arteries pass through the foramen. As well, the thalamostriate, superior choroidal, and septal veins join into the internal cerebral veins on the roof of the third ventricle (Torres-Corzo et al., 2016). The foramen of Monro is a common site for ventricular obstruction. The endoscope is passed through the foramen of Monro during ETV and contains the eloquent memory structure of the fornix which can become damaged. The fornix is a part of the Papez circuit and important for episodic memory (Senova et al., 2020). The fornix has lateralized function with the left fornix responsible

29

for verbal memory and the right fornix for visual memory. ETV are commonly conducted through a right sided approach putting the visuospatial domain at risk. This highlights the need for domain specific neurocognitive testing.

Third Ventricle

This midline cavity is the location where the ETV occurs. The endoscope enters superioranteriorly through the foramen of Monro. The anterior commissure crosses midline anterior to the columns of the fornix and foramen of Monro and is located antero-inferior to the globus pallidus (Lavrador et al., 2019). It contains three divisions into the anterior crus containing olfactory fibers and the posterior crus containing a temporal and occipital portion which may be involved in non-visual communication (Larador et al., 2018; Gerrish et al., 2014). Continuing anteriorly, the lamina terminalis forms the most anterior aspect of the third ventricle. It forms a thin sheet from the optic chiasm to the rostrum of the corpus callosum (Standring, 2008). Fenestration of the lamina terminalis has been used as a surgical corridor for tumour resection, anterior communicating artery clipping, or as an alternative when ETV cannot be performed without obvious functional deficits (Silva et al., 2013; Mao et al., 2019; Giussani et al., 2017). Lastly, the optic recess is formed as a crevice superio-anterior to the optic chiasm.

The roof of the third ventricle commences at the foramen of Monro and continues posteriorly to the suprapineal recess (Standring, 2008). Interhemispheric transcallosal approaches can be used to enter the third ventricle through the four-layer roof: neuronal fornix, two layers of telacoroidea membrane, and blood vessels (medial posterior choroidal arteries and the internal cerebral veins) (Greenwood, 1949).

The floor of the third ventricle is sectioned into two parts: diencephalon and midbrain (Standring, 2008). The most anterior aspect of the floor is the infundibulum, then tuber cinereum, mammillary bodies, posterior perforated substance, followed by medial cerebral peduncles and superior tegmentum. The tuber cinereum is the location where the ETV is made (Torres-Corzo et al., 2016). Once a sufficiently large opening is achieved, CSF will preferentially flow into the preportine cistern bypassing the distal obstruction. An additional layer of arachnoid membrane (Lilliquist membrane) must be dissected through so the CSF can freely flow through the ventriculostomy. The infundibulum and mammillary bodies are millimeters anterior and posterior to the ETV. Although the risk of injury is low, reported cases of pituitary hormonal dysfunction and visual memory deficits have been recorded post-ETV (Bouras & Sgouros, 2011; Bouras & Sgouros, 2013; Benabarre et al., 2001). Mammillary body function is divided by medial and lateral aspects in each body (Tanaka et al., 2020). The medial aspect is responsible for spatial memory and the lateral aspect is responsible for spatial navigation as it contains headdirection cells. The medial aspect is most at risk during ETV again highlight the need for visuospatial domain testing in these patients.

The posterior wall of the third ventricle contains five components from superior to inferior: superpineal recess, habenular commissure, pineal body and recess, and posterior commissure (Standring, 2008). The habenular complex consists of a nucleus on each side of the commissure and is thought to play a role in addiction (Lavrador et al., 2019; Fakhoury et al., 2014). The pineal body is responsible for melatonin release to regulate circadian rhythm (Nichols, 2017). The posterior commissure mediates consensual pupillary light reflex (Lavrador et al., 2019). The lateral walls of the third ventricle consist of the hypothalamus and thalamus with delineation at line with the foramen of Monro via the hypothalamic sulcus (Standring, 2008). Lastly, the massa intermedia/interthalamic adhesion crosses the third ventricle. It contains neurons cell bodies and tracts but is thought to have no or minimal functional significance (Standring, 2008; Damle et al., 2017; Borghei et al., 2020).

Cerebral Aqueduct

This narrow canal connects the third and fourth ventricles (Standring, 2008). It lies between the tectum and tegmentum. It is a common area of obstruction causing aqueductal stenosis (if narrowed) or aqueductal atresia (if absent). Previous attempts at aquductplasty and stenting the aqueduct have been attempted, however the re-occlusion rate is as high as 50% with aquductplasty alone and due to the eloquence of this area, complications include dysconjugate gaze, as well as stent migration, infection, ventriculitis, and subdural hygroma (Torres-Corzo et al., 2016). No evidence shows any benefit to aquductplasty and stenting over ETV.

Fourth Ventricle

The roof of the fourth ventricle consists of the superior and inferior medullary vela, which separate the ventricles from the cerebellum posteriorly (Standring, 2008). The floor is termed the fossa rhomboidei and consists of the posterior aspect of the tegmentum. The fourth ventricle opens into the two lateral recesses of Luschka into the cerebropontine angle and one medial recess of Magendie leading to the cisterna magna. The most inferior aspect of the fourth ventricle is at the level of the obex, where the open medulla oblongata closes and transitions into the

cervical spinal cord. The fourth ventricle narrows into the central canal at this stage. Possible obstructions can result along the fourth ventricle or one of the recesses (termed fourth ventricular outlet syndrome). Obstructions posterior to the tuber cinereum including fourth ventricular obstructions and outlet syndromes meet the anatomical requirements for an ETV to be attempted.

1.5 Surgical Treatments of Obstructive Hydrocephalus: CSF Shunting vs ETV

Obstructive hydrocephalus is unique in that it has two different surgical treatment options, ETV and CSF shunting. CSF shunting actively drains the CSF out of the cerebral ventricles to another location for absorption, typically the peritoneum (Reddy et al., 2014). Although the pleural cavity (Craven et al., 2017), cardiac atrium (Borgbjerg et al., 1998), gallbladder (Demetriades et al., 2013), venous system (Ayaz et al., 2010), and urinary bladder (Ibrahim, 2014) have been used. Lumbo-peritoneal shunting is also often used for iNPH and idiopathic intracranial hypertension (Kazui et al., 2015; Krishnan et al., 2020).

For more than seven decades CSF shunting in hydrocephalus has been shown to be effective at treating neurological and cognitive symptoms of hydrocephalus (Nulsen & Spitz, 1951; Laurence, 1969). However, shunting is also associated with non-trivial infection and blockage rates leading to reoperation for shunt revision. The mean number of shunt revisions per adult patient is 0.6, with 32.5% of total patients requiring revisions, and comprising 39.3% of all shunt surgeries (Reddy et al., 2014). This stratifies into 19.8% of patients requiring a single shunt revision, and 12.7% requiring two or more revisions. Wu et al. (2019) retrospective review of complication post-shunt for iNPH found an overall complication rate of 19% from: misplaced proximal catheters (5%), asymptomatic catheter tract hemorrhages (3%), bilateral hygromas

(4%), subdural hematomas (6%), CSF leak (1%), repeat surgery due to complication (3%). In a European multicenter study using programmable valves, the one-year post-shunt overall complication, subdural hematoma, and infection rates requiring surgery were found in 15%, 1%, and 1% of iNPH patients (Klinge et al., 2012). An additional 13% of patients had complications that did not require surgery, of which 5% were subdural hematomas.

ETV is a minimally invasive bypass procedure using a flexible or rigid endoscopy to perforate the tuber cinereum at the base of the third ventricle. This allows CSF to preferentially flow through the ventriculostomy into the open cisternal spaces creating a pathway around the obstruction to allow for restoration of homeostatic CSF absorption.

ETV is effective at relieving symptoms of raised ICP in adult obstructive hydrocephalus (Isaacs et al., 2016; Waqar et al., 2016). The reported success rate for primary ETV in adults with untreated obstructive hydrocephalus, defined as not needing a subsequent shunt or repeat ETV, ranges from 73% to 87% 10 years post-ETV (Isaacs et al., 2016; Waqar et al., 2016). ETV failure predominately occurs early, with 78% of failures seen within six months after surgery (Dusick et al., 2008; Labidi et al., 2015). It is uncommon for a primary ETV to fail after two years (Isaacs et al., 2016; Waqar et al., 2016; Waqar et al., 2016).

ETV complications are largely transient having an overall morbidity rate of 8.5% (range: 0-31.2%), but permanent morbidity of 2.38% (Bouras & Sgouros, 2011; Bouras & Sgouros, 2013; Kawsar et al., 2015; Schroeder et al., 2002). The overall mortality rate is 0.28-1.28% but is rarely reported in the literature (Jung et al., 2017). The five most commonly reported complication categories include: cognitive, vascular and neurological injury, hemodynamic alterations, endocrinologic abnormalities, electrolyte imbalances, cerebrospinal fluid leakage, fever and infection.

Cognitive complications after neuroendoscopy has also been recently summarized in a systematic review finding transient and permanent cognitive complications in 2.0% and 1.04% in 2804 neuroendoscopy procedures (Soleman et al., 2020). Although the complication rates are relatively low, three of the initial publications about ETV illustrate the severity of the complications that can occur, including: organic personality disorder, immediate memory and executive function impairment from MRI confirmed fornix and mammillary body damage (Benabarre et al., 2001); severe psychotic depression (van Aslst et al., 2002); and permanent episodic memory impairment and bulimia due to right fornix column complete section and hypothalamic lesion to the left wall of the third ventricle (Bonanni et al., 2004).

The major vascular injury rate is 0.49% due to perforation of the basilar artery and its perforators, or septal and thalamostriate veins (Kawsar et al., 2015). Permanent neurological morbidity ranges from 0.5-1.4% (Bouras & Sgouros, 2011; Bouras & Sgouros, 2013; Kulkarni et al., 2016).

Changes in ICP from aggressive irrigation and during ventriculostomy can cause intraoperative hemodynamic changes that are common and largely transient (Kawsar et al., 2015). For example, bradycardia during ventriculostomy stage is reported between 26.8-43.0% (Baykan et al., 2005; Kawsar et al., 2015). However, case reports of intraoperative cardiac arrest and post-operative

uncontrollable high-frequency tachypnea have been reported (Bernard et al., 2010; Handler et al., 1994).

Hormonal changes due to hypothalamic and pituitary injury have an overall morbidity of 0.94% (Bouras & Sgouros, 2011; Bouras & Sgouros, 2013). Diabetes insipidus, weight gain, and precocious puberty have been reported in 0.64%, 0.27%, and 0.04% in pediatric cases (Bouras & Sgouros, 2011). Electrolyte disorders may occur transiently post-ETV (Anandh et al., 2002; Bouras & Sgouros, 2011; Kawsar et al., 2015). Transient post-operative fever is also common and typically resolves within three days (de Kunder et al., 2016; Kinoshita et al., 2013).

Pediatric CSF leak has an overall reported rate of 1.7% and pediatric intracranial infections occur in 1.81-6.1% of cases (Bouras & Sgouros, 2011; Bouras & Sgouros, 2013).

ETV may represent a superior treatment to shunt procedure in adults with chronic obstructive hydrocephalus due to higher patency rate (73-87% vs 67.5%) and lower frequency of permanent complications requiring surgery (2.38% vs 15%). However, the paucity of literature on the efficacy of ETV to improve chronic hydrocephalic symptoms (cognitive, gait, and urinary dysfunction) limits comparability. If it can be demonstrated that ETV improves these chronic symptoms as effectively as shunting, it would make ETV the preferred initial option to treat obstructive hydrocephalus.

1.6 Testing Cognition in Obstructive Hydrocephalus

Cognitive testing can vary greatly by the level of detail required for the clinical assessment or research question. Short ten-minute global assessments like the Montreal Cognitive Assessment (MoCA; Nasreddine et al., 2005) or Mini Mental Status Exam (MMSE; Folstein et al., 1975) can be used to determine whether there exists mild cognitive impairment. However, only the global score that is made up of all the individual tests has clinical validity (Bruijnen et al., 2020). This essentially allows for a binary outcome, either cognitive impairment. But, without additional empirical categories this limits these global assessments to provide more detailed insight into the cognitive abilities of the patient. Additionally, the MMSE is not considered a very sensitive cognitive test to measure the pattern of cognitive deficits observed in hydrocephalus (Golomb et al., 1994; Iddon et al., 1999).

Using more extensive cognitive testing that is still able to be performed at the bed side is crucial for inpatient assessment and in outpatient clinics where detailed full neurocognitive assessment is not possible due to time restraints or when the patient is unable to complete lengthy multi-hour assessments. The Repeatable Battery for the Assessment of Neuropsychological Status revised (RBANS; Randolph et al., 1998) was designed specifically with these constraints in mind. It provides a portable, 20-30-minute assessment with a global cognition score made up of five individually validated cognitive domains: Immediate Memory, Visuospatial and Constructional, Language, Attention, and Delayed Memory. The RBANS allows for comparison to the normal population with percentile rankings from age-grouped normal populations every 8-12 years starting at age 12. As well, there are multiple validated versions to allow for repeated measures

37

testing with high test-retest reliability. This allows for greater interpretation of the global cognition and domain scores.

The greatest level of detail is only achievable with a trained neuropsychologist at an outpatient clinic, typically outside of the initial neurosurgical clinic or emergency room. A full cognitive assessment can take between 3-4 hours or even longer in some cases. As a result, this limits patient populations that are unable to attend additional clinics or are unable to undergo prolonged testing. Amongst the many benefits that do exist with detailed cognitive testing, the thorough nature of many different tests allows for a more holistic understanding of the patients cognitive and behavioural capacities, which can allow for discrimination between psychiatric and neurological symptoms. In addition, extensive domain specific testing allows for the most accurate and specific domain abilities and allows for changes over time to be easily tracked with clinically meaningful changes often available for interpretation.

Based on the chronic obstructive hydrocephalus patient population being studied in this thesis, it is essential that portable and relativity short assessments are selected able to be completed. The MoCA allows for quick and easy global assessment and the RBANS allows for a subset of patients to undergo more detailed testing at the bed side and is within the limited testing length this population is able to endure.

1.7 Evidence of ETV to Improve Cognition in Obstructive Hydrocephalus

The study of cognitive changes in chronic obstructive hydrocephalus treated with ETV is quite limited. Modern ETV was first described in 1978 by Vries and later made popular by Jones et al.

(1990). Teo (1998) reported 13 patients underwent neuropsychological testing but did not publish the results. Larsson et al. (1999) reported two patients with unrecognized congenital aqueductal stenosis had some improvement based on their overall symptom score. However, whether there existed any cognitive change post-ETV is not possible to determine based on the reported data.

The first paper on cognitive outcomes post-ETV was not published until Burtscher et al. (2003) provided pre- and post-ETV detailed cognitive testing in six adult patients with unrecognised congenital aqueductal stenosis. All six patients improved in multiple cognitive tests by means of individual analysis with descriptive statistics. Despite this being the first paper on cognitive outcomes, there was not another study using pre- and post-ETV cognitive assessments until Locatelli et al. (2014) evaluated eight patients with unrecognized congenital aqueductal stenosis and pre-ETV cognitive deficit pre- and post-ETV. They found three patients had improved and two patients had complete resolution of their cognitive deficit. However, the method of assessment was not reported, nor was there any indication of objective cognitive evaluation. Hader et al. (2014) used neuropsychology assessments evaluated with a Reliable Change Index in five pediatric and eight adult patients with chronic obstructive hydrocephalus pre- and post-ETV. Nine of 13 patients demonstrated improvement in at least one of six cognitive domains tested. Baroncini et al. (2019) used the MMSE in 15 patients with chronic obstructive hydrocephalus and found no significant difference in mean MMSE scores post-ETV. However, the MMSE is not considered a very sensitive cognitive test to measure the pattern of cognitive deficits observed in hydrocephalus (Golomb et al., 1994; Iddon et al., 1999).

Hamada et al., (2009), provided a case report of an unrecognized congenital aqueduct stenosis with normal pre-ETV scores on the MMSE (29/30), an intelligence quotient of 115, and normal RBANS domains, except visuospatial and constructional domains which was just outside of normal (84; 85-115 normal range). Post-ETV all cognitive domains significantly improved. This case report provided the first evidence against the idea that normal cognition pre-ETV excludes possible improvement post-ETV. The second evidence came from Hong et al. (2016), who presented a case report of unrecognized congenital aqueduct stenosis found incidentally and remained asymptomatic for ten years and underwent baseline neuropsychological testing. Thereafter, the patient was treated with ETV for increasing headache, memory loss, gait instability, and urinary and fecal incontinence. 15-month post-ETV neuropsychological testing had significantly improved compared to the pre-ETV baseline.

Lacy et al., (2009) conducted post-ETV cognitive assessments in ten patients with obstructive hydrocephalus. The assessments included the RBANS showing 40% of patients had >2SD cognitive deficit in two or more domains, 50% of patients had emotional testing suggestive of depression, and 30% of patients had anxiety-related symptoms. They noted specific deficits in processing speed, response inhibition, cognitive flexibility, and verbal and visual memory. However, intelligence quotient was found to be normal (study mean = 97; normal = 100 with SD =15) using the Weschsler Test of Adult Reading. Al-Jumaily et al. (2012) also used the RBANS in a group of twenty unrecognized congenital obstructive hydrocephalus post-ETV to determine how they compared to the normal population. The mean RBANS total score and domains scores were all below the normal range. Xhu et al., (2013) found seven of nine adults who underwent

ETV and biopsy have post-operative normal cognitive scores on the MMSE and Rivermead Behavioural Memory Test.

This relative lack of studies that assess cognition in a systematic prospective manner, both preand post-ETV, has limited our ability to quantitatively assess both the potential cognitive benefits and risks of ETV. This provides weak and conflicting evidence of cognitive change post-ETV due to their heterogeneous results, small sample size, and variation in cognitive testing.

1.8 Research Question

To address the gaps in the literature, we proposed to use neurocognitive assessments pre- and post-ETV in prospective cohorts to demonstrate how global cognition and multiple cognitive domains are affected by ETV treatment of chronic obstructive hydrocephalus in adults.

We hypothesize that ETV will improve global cognition based on the existing literature and expert opinion. Hydrocephalus is thought to follow a frontal-subcortical dementia pattern therefore we hypothesize that ETV will improve immediate memory, delayed memory, and attention domains, but not the cortically driven language domain. Additionally, ETV may cause damage to visuospatial networks as has been described in the literature and based on surgical manipulation around important anatomy to these systems.

Using this framework, this thesis will meet the following specific objectives:

O1. To determine the sufficiency of primary ETV to safely improve global cognition in adults with chronic obstructive hydrocephalus. We will use the MoCA to measure global cognitive changes pre-ETV and twice post-ETV prospectively in a multicenter study.

O2. To determine which cognitive domains ETV produces clinically meaningful changes in adults with chronic obstructive hydrocephalus. We will use the RBANS to individually measure five cognitive domains pre-ETV and once post-ETV in a prospective single center study.

Chapter 2: Short- and Long-term Follow-up of Cognitive Outcomes after Primary ETV in Adult Chronic Obstructive Hydrocephalus

The first objective of this thesis is to determine if ETV safely improves cognition in adult patients with chronic obstructive hydrocephalus. The MoCA and SDMT allows for assessment of global cognition, processing speed and task switching with short assessment time. This will allow for multi-center data collection within the Adult Hydrocephalus Clinical Research Network (AHCRN). Results from this paper will provide the first prospective data, solely in adults with both pre- and post-ETV testing over three time points.

2.1 Methodology

The details of the AHCRN registry and its methods have been previously described (Williams et al., 2019). Subjects were enrolled prospectively; however, not all patients seen at each center could be enrolled.

2.1.1 Inclusion and Exclusion Criteria

Subjects for this study enrolled in the AHCRN registry between December 2014 and January 2019 and had primary ETV procedures between April 2015 and January 2019, with a minimum of three months follow-up after the procedure. From subjects in the Unrecognized Congenital and Acquired categories, we identified those with obstructive hydrocephalus, potentially amenable to ETV, based on brain CT or MRI with evidence of obstruction to CSF flow at the cerebral aqueduct, within the 4th ventricle, or at the foraminae of Luschka and/or Magendie.

The purpose of this study was to examine the effect of primary ETV on cognition in adults with chronic obstructive hydrocephalus. Therefore, subjects were excluded if: 1) The hydrocephalus had previously been treated either with an ETV or a shunt; 2) Pre- and post-ETV MoCA was unavailable; or 3) The patient required a subsequent shunt procedure or repeat ETV, within five months of the primary ETV.

2.1.2 **Baseline Evaluations**

MoCA versions 7.1, 7.2, and 7.3 (Nasreddine et al., 2005) and SDMT original version, alternative 1, and alternative 2 were used to assess global cognition and processing speed (Smith, 1982; Benedict et al., 2017). The MoCA contains 12 items that fit into seven cognitive domains and sum into a total score between 0-30. A total score of greater than 25 is considered normal. Although the total score has good psychometric properties, the cognitive domains should be interpreted with caution (Bruijnen et al., 2020). Therefore, only the MoCA total score (MoCA TS) was used in this study. A change in MoCA TS of \geq 2 points was defined as clinically significant (Krishnan et al., 2017). A clinically significant change in SDMT was defined as \geq 4 points (Smith, 1982; Benedict et al., 2017). Only primary ETV patients that had documentation of a pre- and post-ETV MoCA were included in the analysis.

2.1.3 Intraoperative Data

Data recorded included: type and size of endoscope used, bleeding during procedure, any surgical parenchymal contusion, vascular/vessel injury, if a biopsy was performed, if the procedure was abandoned, the method and instrument used to create and enlarge the opening in the floor of the third ventricle, any additional procedure performed, CSF appearance at the

beginning and end of the procedure, whether an open pre-pontine cisternal space was achieved with basilar visualization, estimated size of fenestration and if an EVD was inserted. Any additional intra-operative or subsequent post-operative complications were also recorded.

2.1.4 Statistical Analysis

Binary and categorical data are presented as numbers and percentages. Continuous outcomes, which tended to have skewed and/or long-tailed distributions, are presented using medians and the interquartile range (IQR). IQRs are presented using the notation [X-Y], indicating that approximately one-quarter of observations have values less than X, one-quarter have values greater than Y, while the middle 50% of observations lie in the interval [X-Y]. The within-patient change medians for MoCA TS were determined by subtracting the pre-ETV score from the post-ETV score of each patient and taking the median of the resulting change scores. Assessments of whether within-patient change scores differ significantly from zero were performed using the Wilcoxon signed rank test. All reported *p*-values are two-sided, with a significance level of $\alpha = 0.05$.

2.2 Results

2.2.1 Patient Population (Figure 1)

A total of 74 patients underwent primary ETV, of which 38 patients were analyzed. Thirty-six ETV patients were excluded from this study as they could not complete either a pre- and/or post-ETV MoCA assessment. The most common reasons for exclusion included drowsiness or cognitive deficits that precluded their ability to follow instructions to complete cognitive assessments. However, this group of 36 patients was still followed and assessed at four months post-ETV. Compared to pre-ETV, 12/36 could complete a MoCA test post-ETV. Therefore, many of these 36 excluded ETV patients did demonstrate substantial improvement in cognition post-ETV, but this improvement could not be measured due to the lack of objective pre-ETV data.

Of the 38 patients, 17 (45%) were female. The mean age was 51.9 ± 17.1 years. Most patients were Caucasian (n=33; 87%), with the remainder Asian.

The most common hydrocephalus category was Unrecognized Congenital (n=28, 74%), which included aqueduct stenosis (n=21, 55%), aqueductal pattern (n=6, 16%) and arachnoid cyst (n=1, 3%). The remaining 10 patients (26%) were in the Acquired category, which included brain tumour (n=6, 16%; 2 malignant and 4 benign), intraventricular adhesion or web (n=3, 8%), and other masses or vascular anomalies (n=1, 3%).

Pre-operative symptoms included: cognitive problems (n=26, 68%), gait disturbance (n=22, 58%), headache (n=19, 50.0%), and urinary dysfunction (n=18, 47%). Nineteen patients (50.0%) presented with combined dysfunction of cognition, gait, and urination.

An open pre-pontine cisternal space with basilar visualization was recorded by the surgeon in 33 procedures (87%). In the remaining five patients, the ETV was felt to be successful based on improvement of baseline signs and symptoms, decreased ventricular size on post-ETV imaging and absence of the patient requiring an additional CSF diversion procedure within five months of the primary ETV.

The mean early post-ETV assessment was at 4.6 ± 4.0 months (n=38) and is labelled as "5-Months Follow-up" (Tables 1 & 2; Figures 1, 2 & 3). The mean late post-ETV assessment was at 14 ± 3.1 months (n=15) and is labelled as "14-Months Follow-up".

2.2.2 MoCA (Table 1 and Figure 2)

<u>5-Months Follow-up (n=38)</u>: The baseline median MoCA TS score was 24/30 [23-27]. The 5months follow-up median MoCA TS score was 26/30 [24-28]. The median within patient change was +1 point [0-2], p < 0.001, which does not meet the threshold of ≥ 2 points for clinical significance (Krishnan et al., 2017).

<u>14-Months Follow-up (n=15)</u>: The baseline median MoCA TS was 23/30 [22-27]. The 14months follow-up MoCA TS was 26/30 [25-28]. The median within patient change was +2 points [1-3], p = 0.007, which is also clinically significant.

2.2.3 SDMT (Table 2)

The SDMT scores did not change following ETV at either the 5-months or 14-months follow-up (Table 2). For the 5-Months follow-up (n=32), the baseline median SDMT score was 39 [29-48], 5-months follow-up median was 45 [35-50], and the median within patient change was +3 [-1-6], p = 0.084. For the 14-Months follow-up (n=14), the baseline median SDMT score was 32 [28-43], the 14-months follow-up median score was 38 [30-46], and the within patient change was +3.0 [1.0-8.0], p = 0.138. Both the 5-months and 14-months follow-up change scores failed to meet the clinical significance criterion of ≥ 4 points (Smith, 1982; Benedict et al., 2017).

2.3 Discussion

Our prospective and multicenter study shows that when adults with untreated chronic obstructive hydrocephalus present with cognitive impairment, treatment with ETV results in significant and sustained improvement for as long as 14 months. Although, early clinically significant improvements were not seen at 5-months follow-up. This suggests that longer follow-up is needed to identify this gradual cognitive improvement.

Our study provides the largest dataset to date on cognitive impairment at the time of presentation in adults with chronic obstructive hydrocephalus, as well as its response to ETV. This relative lack of studies that assess cognition in a systematic manner, both pre- and post-neuroendoscopy (Soleman & Guzman, 2020), has limited our ability to quantitatively assess both the potential cognitive benefits and risks of neuroendoscopy. Only four relevant cohort studies have included pre- and post-ETV cognitive assessment (Burtscher et al., 2003; Locatelli et al., 2014; Hader et al., 2014; Baroncini et al., 2019). One study stated that 5/8 patients, who had pre-ETV cognitive impairment, improved post-ETV but the method of assessment was not provided, and an objective evaluation of change was not reported (Locatelli et al., 2014). Nine of 13 adult and pediatric patients demonstrated improvement in at least one of six cognitive domains using neuropsychology assessments evaluated with a Reliable Change Index (Hader et al., 2014). Burtscher et al. (2003) found 6/6 patients demonstrated some cognitive improvement post-ETV, utilizing a large battery of cognitive tests, however only used descriptive statistics and individual analysis. Baroncini et al. (2019) found no significant difference in mean MMSE scores post-ETV in 15 patients. However, the MMSE is not considered a very sensitive cognitive test to measure

the pattern of cognitive deficits observed in hydrocephalus (Golomb et al., 1994; Iddon et al., 1999). These studies provide weak and conflicting evidence of cognitive change post-ETV due to their heterogeneous results, small sample size, and variation in cognitive testing.

In our study, the change in median MoCA score at the 5-months follow-up was overall not clinically significant, although 40% of patients did have an increase score of at least two points. This could be due to insufficient power or subgroup differences. However, the improvement at 14-months follow-up was clinically significant and indicates that treating chronic obstructive hydrocephalus with ETV results in long-term improvement in cognition, although 40% of the group had no change and one patient (7%) worsened. Increased numbers in our long-term cohort will improve the validity of any cognitive changes seen in chronic obstructive hydrocephalus patients post-ETV.

Cognitive processing speed and efficiency, as measured by SDMT, was not improved at either the 5-months or 14-months follow-up. This test has been shown to be sensitive to processing speed in the multiple sclerosis population (Benedict et al., 2017). Detailed cognitive assessment with a comprehensive neuropsychological battery should be carried out in the future.

In summary, statistically significant but not clinically significant improvement in cognition may not be evident at 5-months post-ETV but becomes both clinically and statistically significant by one-year post-ETV. In addition, our data demonstrate that ETV can be performed safely with a low rate of cognitive deterioration: worsening of global cognition was rare, both at 5-months (2 patients) and at 14-months (1 patient) following ETV. Chapter 3: Improved Global Cognition, Attention, and Delayed Memory After Endoscopic Third Ventriculostomy in Adult Chronic Obstructive Hydrocephalus Using The Repeatable Battery for the Assessment of Neuropsychological Status (RBANS)

The second objective of this thesis was to determine which cognitive domains are affected after ETV for adult chronic obstructive hydrocephalus. The RBANS allows for assessment of global cognition and five major cognitive domains: immediate memory, visuospatial/constructional, language, attention, and delayed memory. This will allow for determination of which domains of cognition are causing the global changes seen with the MoCA. Additionally, a novel pre-ETV predictor of post-ETV global cognition improvement will be discussed.

3.1 Methodology

3.1.1 Inclusion and Exclusion Criteria

Subjects were enrolled prospectively as they presented to our outpatient clinic or emergency department between April 2018 and March 2020, with a minimum of three months follow-up after the procedure. We identified those with obstructive hydrocephalus, potentially amenable to ETV or endoscopic resection, based on brain MRI with evidence of obstruction to CSF flow at the foramen of Monroe, cerebral aqueduct, within the 4th ventricle or at the foraminae of Luschka and/or Magendie. Subjects were excluded if they could not complete pre- or post-ETV assessments or if the patient required subsequent resection, shunt procedure, or repeat ETV within five months.

3.1.2 Assessments

The patient underwent the following assessments pre-ETV and once post-ETV. The MoCA is a five- to ten-minute global cognitive assessment (Nasreddine et al., 2005). It is based on 12 items that fit into seven cognitive domains and sum into a total score between 0-30. A total score greater than 25 is considered normal. Although the total score has good psychometric properties, the cognitive domains should be interpreted with caution (Bruijnen et al., 2020). Therefore, we have elected to only consider the total score in our analysis. A change in MoCA total score (MoCA TS) of \geq 2 points was defined as clinically significant (Krishnan et al., 2017). MoCA versions 7.1, 7.2, and 7.3 were used.

The RBANS is a 25-minute neurocognitive assessment (Randolph et al., 1998). It is based on 12 subtests that fits into five cognitive domains (immediate memory, visuospatial and constructional, language, attention, and delayed memory) and sum into a total score for global cognition with percentile. The total score and indexes are based on age-grouped normal populations with a mean of 100 and standard deviation of 15. We elected not to determine whether our population was significantly different from the normal population, as the purpose of this study was to establish initial data on how cognitive domains change within patient post-ETV.

There is exist some controversy on the psychometric properties of the original domain groupings, with many suggesting new grouping of the subtest (Wild et al., 2006; Duff et al., 2006; Carlozzi et al., 2008; Garcia et al., 2008; Yang et al., 2009; Duff et al., 2010; Schmitt et al., 2010; King et al., 2012; Torrence et al., 2016; Vogt et al., 2017). But, many of these studies were conducted on

mixed group or normal populations, which leads to ceiling effects during testing. Two recent papers that assessed the RBANS on Alzheimer's and suspected cognitive impairment patients found good psychometric proprieties in the five original domains (Emmert et al., 2018; Holden et al., 2020). Therefore, we selected to keep the original five domains as well as the total score as our primary outcome variables. Secondary analysis included the 12 subtests and the effect of age.

There exist neither a perfect Minimum Clinically Important Difference (MCID) or Reliable Change Index (RCI) for the RBANS that fits our sample's needs. Phillips et al. (2015) MCID has been critiqued for design compared to the available RCI (O'Connell,2019). However, the cohorts were not equal in cognitive performance (healthy controls vs dementia) and all domains except for the visuospatial/constructional domain MCIDs were satisfactory. Nonetheless, careful interpretation of the visuospatial/constructional domain should be utilized, and subtest interpretation can be used for more precise interpretation of subdomain changes. In contrast, the RCI available is based on an elderly (>65 years old) population and is less robust at percentile extremes (Duff, 2004 & Duff, 2005). Since our sample's mean age is 34 and the percentiles for the primary outcomes are extremely low, the MCID was selected as the clinically changed cutoff. This will allow for a general idea of how the sample changed clinically. Although further validation of the MCIDs and RCIs of RBANS on hydrocephalus patients is required for more robust results. The MCID are as follows: RBANS TS \geq 8, Attention \geq 4, Delayed Memory \geq 10, Language \geq 9, Immediate Memory \geq 10, Visuospatial/Constructional \geq 6.

3.1.3 Intraoperative Data

Data recorded included: type and size of endoscope used, bleeding during procedure, any surgical parenchymal contusion, vascular/vessel injury, if a biopsy was performed, if the procedure was abandoned, the method and instrument used to create and enlarge the opening in the floor of the third ventricle, any additional procedure performed, CSF appearance at the beginning and end of the procedure, whether an open pre-pontine cisternal space was achieved with basilar visualization, estimated size of fenestration, and if an EVD was inserted. Any additional intra-operative or subsequent post-operative complications were also recorded.

3.1.4 Statistical Analysis

Binary and categorical data are presented as numbers and percentages. Continuous outcomes with skewed and/or long-tailed distributions, are presented using medians and the interquartile range (IQR). IQRs are presented using the notation [X-Y], indicating that approximately one-quarter of observations have values less than X, one-quarter have values greater than Y, while the middle 50% of observations lie in the interval [X-Y]. Otherwise, 95% confidence interval are displayed as (lower bound, upper bound). The within-patient change medians were determined by subtracting the pre-ETV score from the post-ETV score of each patient and taking the median of the resulting change scores. Assessments of whether within-patient change scores differ significantly from zero were performed using the Wilcoxon signed rank test. All reported *p*-values are two-sided, with a significance level of $\alpha = 0.05$. Correlations were computed with Pearson's r. No familywise correction was used due to the exploratory nature of the study design.

3.2 Results

3.2.1 Patient Population (Figure 3)

A total of 29 ETV candidates underwent pre-ETV RBANS testing, of which 17 underwent went primary or secondary ETV, with one patient receiving an endoscopic resection without ventriculostomy. Ten patients completed the follow-up assessment, of which seven primary ETVs were performed, two secondary ETVs and one endoscopic resection without ventriculostomy. Twenty ETV patients were excluded from this study as they did not undergo an ETV or did not complete any post-ETV testing. The most common reasons for exclusion included: incomplete testing either pre-ETV, lack of symptomology to indicate surgery (n=4), and ETV requiring subsequent diversion procedure before post-ETV assessment (n=4). However, this group of 20 patients is still being followed for standard clinical follow-up.

Of the ten patients analyzed, seven were female. The mean age was 34.00 years (SD: \pm 16.98, range: 17-68). Patients were Caucasian (n=9) or Asian (n=1). The most common hydrocephalus category was the acquired category (n=6), which included brain tumour (n=3, 1 benign midbrain tectal glioma, 1 benign thalamic tumour, 1 hemangioblastoma), colloid cyst (n=1), pineal cyst (n=1), and cavernous malformation (n=1). Two patients (20%) were in the unrecognized congenital category, which included aqueduct stenosis (n=1) and arachnoid cyst (n=1). Two secondary ETVs were conducted on etiologies of previously shunted aqueductal stenosis (n=2).

Pre-operative symptoms included: decreased level of consciousness (n=2), headache (n=7), nausea and vomiting (n=2), and cognitive problems (n=6).

An open pre-pontine cisternal space with basilar visualization was recorded by the surgeon in all nine ETVs. The endoscopic resection of a 3rd ventricular colloid cyst was felt to be successful based on successful near total resection, improvement of baseline symptoms and signs, decreased ventricular size on post-operative imaging, and absence of the patient requiring an additional CSF diversion procedure within 3-months.

Pre-ETV assessment was at 22.6 ± 44.5 days before ETV. The follow-up assessment was at 3.8 ± 2.3 months and is labelled as 4-Month Follow-up (Table 2).

3.2.2 MoCA (Table 2 and Figure 4)

The 4-month (n=9) follow-up median within patient change in MoCA TS was not significantly different, 0 point (-2, +1), p = 0.7204 (Table 2; Figure 6). The baseline and post-ETV median MoCA TS were 23/30 and 24/30, which is outside the normal range of $\ge 26/30$. At baseline, 2/9 patients had MoCA TS in the normal range. Post-ETV only one patient had a normal score. Individual clinical change shows 2/9 patients had a clinically significant improvement and 3/9 patients had clinically significant worsening of ≥ 2 points (Krishnan et al., 2016).

3.2.3 **RBANS (Table 2 and Figure 5)**

In contrasts to the MoCA data, RBANS Total Score (RBANS TS) post-ETV was significantly improved with median a within patient change of +8.0 [+4.25, +15.50], p = 0.0125. This also met the MCID criteria of \geq 8 as being clinically improved. Individually, 50% (5/10) showing clinically significant differences in total score and no significant worsening. However, the

median RBANS TS was still > 1SD below normal (100) at pre-ETV, 72 [55-81], and post-ETV, 77 [66-89].

The Total Score Percentile (TSP) was also significantly improved from a median of 3.5 [0.33-11] to 6.5 [1.25-23.5], with a change score of +3.45 [+0.30, +18.00], p = 0.0284. It should be noted that since the total score was significantly different, the TSP will likely be significant as well since they contain the same data in different unit conversions (ie. colinear). The magnitude of within patient change in TSP was strongly correlated to the pre-ETV TSP, r = 0.83 (0.43, 0.96), p = 0.0027 (Figure 6).

The standard five-factor analysis of the 12 subtests showed that the improvement in global cognition came from the Attention and Delayed Memory Index Scores, p = 0.0058 and p = 0.0143. Both index scores also met their MCIDs. Pre-ETV medians for Attention and Delayed Memory and the Attention post-ETV median were 1 SD below the normal population mean. Delayed Memory was the only domain to improve into the normal range post-ETV. Immediate Memory, Visuospatial/Constructional, and Language were not significantly different, nor did they meet their respective MCIDs. Language was the only other domain whose median scores were within ± 1 SD of normal: Pre-ETV, 88.0 [57.00-95.25]; Post-ETV, 85.5 [76.0,105.0].

The Attention Index Score is a composite of the digit span and coding subtests. Digit span was significantly improved by +1.5 points [0.25, 2.75], p = 0.0397. As subtests do not have half points, interpretation of median changes will be a range from the closest two whole numbers. This equates to an extra one or two second-string repetition on the test or one additional first-

string item out of eight items. This is a median 9.375% improvement on this subtest. The Coding subtests was nearly statistically significant at +5.5 [2.5, 7.5], p = 0.0523. Six additional correct matching is a 6.74% improvement on this test.

The Delayed Memory Index Score is a composite of four subtests using a summed score from three recall subtests (Figure Recall, List Recall, and Story Recall) and the List Recognition subtest. The Figure Recall was significantly improved by +5.0 points [2.5-5.75], p = 0.0049. This is an additional five correct drawing or placement items and a median 22.5% improvement on the subtest. List Recall was trending towards significance at +1 point [0.00, 2.75], p = 0.0790. This represents a 10% improvement in remembering ten items without cueing. Story Recall was not significant, p = 0.2736. However, when summed together there was a statistically significant improvement of +5.5 point [4.25, 11.50], p = 0.0058. The total possible summed score is 42, therefore a median 13% improvement was seen in delayed memory recall. However, List Recognition was not significantly improved, p = 6989. Thus, the Delayed Memory index score interpretation should be limited to improved figure recall and potentially list recall, but not recognition.

Picture naming was significantly improved by +4 [1.0,5.0], p = 0.0162. It is one of the components for the language index score, the other being semantic fluency, which was not significant, p = 0.3571. Story memory, a part of the Immediate Memory index, is the only other subtest to be marginally trending towards significance at p = 0.1000. All other subtests were not significantly different.

3.2.4 Pre-ETV Memory Domains as a Predictor of Post-ETV RBANS TS Improvement

When compared individually, all five of the patients who had clinically improved RBANS TS post-ETV also had similar Immediate and Delayed Memory index scores, within ± 3 points. The Immediate Memory and Delayed Memory Index Scores were subtracted from each other and the difference was plotted against their paired change in RBANS TS (Figure 7). Retrospectively, the memory difference score perfectly separates those who had clinically significant improvement from those who did not. No other pre-ETV indexes were similar to each other to suggest a different predictor.

3.2.5 Individual Analysis

Individual analysis shows that when the MoCA TS and RBANS TS were compared in this small sample, the MoCA TS showed three patients who had clinically significant worsening, but their RBANS TS did not. In fact, one of the three patients had an improved RBANS TS and was the only patient to improve in 4/5 domains. The other two patients had no clinically different RBANS TS. These two patients had 1-3 domains clinically improve and 0-1 clinically worsened domain. Both patients who had a significant improvement in the MoCA TS also had a significant improvement in RBANS total score. Of the four patients who had no significant change in MoCA TS, two had significant improvements in RBANS TS.

RBANS TS and Delayed Memory had 5 patients who met the MCID for clinical improvement and no patient worsened. Attention had seven patients clinically improve and no worsening. Immediate memory had four patients clinically improve and no worsening. Language had five patients clinically improve and one who had clinically worsened. Visuospatial/Constructional
had two patients clinically improve and five who clinically worsened. The patients who had Language worsening also had clinically worse visuospatial/constructional domain.

Overall, all patients had at least one significantly improved domain, seven patients had at least two significantly improved domains, four patients had at least three significantly improved domains, and one had four significantly improved domains. No one improved in all five domains.

3.3 Discussion

This prospective study with multidomain cognitive testing shows that when adults with chronic obstructive hydrocephalus undergo treatment with ETV, clinically and statistically significant improvements in Attention and Delayed Memory domains are present at 4-months follow-up. Global cognitive changes showed conflicting results depending on method of assessment, however more robust testing suggests significant improvement in global cognition. Although not statistically significant, individual analysis revealed the Visuospatial/Constructional domain may be worsened initially post-ETV. Lastly, pre-ETV memory domains may predict post-ETV global cognitive improvement.

3.3.1 MoCA

In this follow-up study to the previous chapter, the median baseline MoCA TS was the similar at 23/30 and 24/30, suggesting that these two samples had comparable pre-ETV cognitive symptoms. The median short-term (4/5-months) follow-up scores were similar, showing no clinically significant change. The percentages of patients that showed clinical improvement

59

differed from 20% (2/10) to 40% (15/38). The striking difference was that this sample had 30% (3/10) patients who clinically worsened versus 5% (2/38) in the previous study. Due to the small sample size it is unclear whether these differences were simply due to sampling error or whether selection bias played a role. The bias for continued follow-up in those who may be still symptomatic could be overrepresented in those who complete the follow-up assessment and are available for analysis. It is likely that the small sample size led to a component of both sampling error and selection bias. Since this is an exploratory study, more conservative estimates of improvement will not lead to exaggerated conclusions. Mindful interpretation of the generalizability is still required.

3.3.2 RBANS TS and TSP

This is the first study on ETV cognitive outcome that used validated multi-domain cognitive testing with good psychometric properties collected prospectively solely in adults pre- and post-ETV. This small study helped to bridge the gap within the limited literature surrounding cognitive changes post-ETV. By using the RBANS TS we provided additional evidence for global cognitive improvement in 50% (5/10) of patients at short-term follow-up. Additionally, the RBANS TS found 0% (0/10) of patients had clinically or statistically significant worsening of global cognitive.

Only four cohort studies had both pre- & post-ETV cognitive testing (Baroncici et al., 2019; Locatelli et al., 2013; Hader et al., 2014; Burtscher et al., 2003). They showed no significant difference, 63%, 69%, and 100% improvement in cognition post-ETV. However, Baraoncici et al., (2019) found used the MMSE, which has been shown to have poor sensitivity in hydrocephalus patients (Golomb et al., 1994; Iddon et al., 1999). Hader et al. (2014) showed that 69% (9/13) mixed-aged patients improved in at least one of six cognitive domains tested, with 17% worsening in executive function. Burtscher et al (2003) by way of descriptive statistics and individual analysis found 100% of patients improved in some of their pre-ETV deficits, with two patients returning to normal scores, three patients with improved but abnormal scores, and one patient with continued deficits in some domains with improvement in others. Locatelli et al. (2014) listed 63% (5/8) patients with pre-ETV cognitive deficit had improved (3/8) or resolved (2/8) symptoms post-ETV. However, the method of assessment was not included and no objective evaluation of change was reported. Similarly, the RBANS TS in our sample had > 1SD deficit in 8/10 patients pre-ETV and 6/10 patients post-ETV. In this view, two patients returned to normal scores. But, by looking at the RBANS TS MCID 50% (5/10) of patients improved in global cognition. Included in those improved patients were the two patients who had normal scores of 99 and 100 pre-ETV, which improved by 9 and 16 points respectively post-ETV. Therefore, normal scores on the RBANS TS does not limit the patient from improving post-ETV. Thorough history and examination should still be done to determine each patient's own normal and any deviations from it.

The Total Score Percentile (TSP) is an alternative representation of the RBANS TS that does not rely on MCID, but instead is compared to an age-matched normal population. The TSP demonstrated the extremely low median global cognitive percentiles this population has before (3.5%) and after surgery (6.5%). Burtscher et al (2003) also found all six patients had multiple tests scores below the 3rd percentile pre-ETV. Hader et al. (2014) also found that 11/12 patients pre-ETV had at least one 'borderline' domain $\leq 16^{th}$ percentile (one patient excluded as four of

61

six domains were not tested) and 8/12 patients had at least one 'impaired' domain $\leq 7^{\text{th}}$ percentile.

Although statistical and clinical improvement in global cognition was seen in the RBANS TS, the median within patient change in TS percentile was only +3.45% with 7/10 patients still under the 10^{th} percentile and 2/10 patients >50^{\text{th}} percentile post-ETV. These are the same two patients with normal RBANS TS. The magnitude of change was strongly correlated with Pre-ETV TSP (r = 0.83). Showing that those with a higher pre-ETV percentile tend to improve the most. This may have implications for treating those with minimal or mild cognitive dysfunction and evidence of obstructive hydrocephalus. A case report found that a 39-year-old woman with a pre-ETV intelligence quotient (IQ) of 115, MMSE =29/30, and RBANS TS and domains within +/-1SD of norms with the exception of visuospatial and constructional, had significant improvement in immediate and delayed memory, and visuospatial and constructional domains (Hamada et al., 2009). Hong et al. (2016) also presented case report of incidental, asymptomatic, unrecognized congenital aqueduct stenosis for normal baseline neuropsychological testing. Ten years later, the patient represented with increasing headache, memory less, gait instability, and urinary and fecal incontinence and was treated with ETV. The 15-month post-ETV neuropsychological testing showed significant improvements compared to the pre-ETV baseline scores. As well, delaying treatment until further cognitive dysfunction may limit the amount of improvement seen post-ETV, as has been shown in the shunting literature (Kazui et al, 2015 (ie. Sinphoni trial); Yamada et al., 2017 (post-hoc Sinphoni trial); and Andren et al., 2014).-In sum, a normal RBANS TS does not exclude patients from improvement as those with the highest pre-ETV scores may see the greatest improvements.

62

3.3.3 Index Scores

Using the standard indexing, attention and delayed memory were statistically and clinically improved at 4-months follow-up. Individual analysis showed that 70% of patients had clinically improved attention and 50% showed clinically improved delayed memory. No one had clinical worsening in these domains. However, both pre- and post-ETV median were below the normal range.

Attention domain improvements post-ETV have been shown previously (Hader et al.,2014). This study found improvement in the attention domain with 33% improving with no worsening post-ETV. Their median pre-ETV attention percentile was 27th but did not reach their definition of borderline or impaired. Improvements in attention post-shunting in iNPH has been previously seen as well (Liu & Su, 2020). However, long-term attention problems may persist more than 20 years after shunt insertion in pediatric patients (Gmeiner et al., 2019). Although there was improvement in attention post-ETV, it is possible that attentional problems may persist throughout life.

The Attention domain is derived from the Digit Span (forwards) subtest and the Coding subtest. The functional implication of improvements in these subtests and therefore the index scores are increased working memory storage capacity (McGrew, 2009) and improved processing speed and task switching (Smith, 1982). Working memory storage capacity can be thought of as the ability to guide behaviour by using and retaining information over short time intervals (Drew & Vogel, 2009; Goldman-Rakic, 1987). The coding subtest is similar to the SDMT used in the previous study. Interestingly, there was a near significant difference in this cohort (p = 0.0523) and a trend towards significance at 4-months using the SDMT in the larger cohort (p = 0.084). Future studies evaluating the long-term attentional outcomes are needed. As well, the impact of attentional improvements on quality of life is warranted.

Delayed memory recall but not recognition was improved in our study. This may be indicative of different networks being affected. Recall has been causally associated with the hippocampalcortical network (Wang et al., 2014; Nilakantan et al., 2019). Recognition is based on two distinct processes, recall and familiarity discrimination (Brown, 2015). The latter has been shown to be found in the perirhinal cortex of the temporal lobe (Brown, 2015). Chronic hydrocephalus cognitive dysfunction typically follows a frontosubcortical dementia pattern (Ogino et al., 2006). This difference in location between subcortical networks and temporal cortex may explain why recall but not recognition is improved.

Visuospatial and constructional index had the largest amount of clinically worsened scores by individual analysis (50%), but was not statistically changed, p = 0.3850. The figure copy subtest had two patients achieve max scores post-ETV and two patients worsen post-ETV (-1 and -3 points). Line orientation subtest had six patients with a negative change score. This suggest that this subtest makes up a large proportion of worsening in the visuospatial/constructional index score. Line orientation is a basic visuospatial task. Worsening scores in 50% of patients regardless of statistical significance is concerning for post-ETV deficits in this domain.

Through lesion-mapping, subcortical fibers associated with performance on line orientation and figure copy have been associated with the right frontal area when the endoscope is advanced through the brain (Biesbroek et al., 2014). However, the lesion-mapping studies produce large areas of fMRI activity and are not conclusive as anatomical correlates. Additionally, the subcortical pathway for visuospatial function is poorly understood. More detailed visuospatial assessments should be performed and compared to left verse right endoscope placement and whether right forniceal and medial mammillary body injury intra-operatively is correlated to these outcomes.

Lastly, although the language domain was found to be not significantly changed, p = 0.2300. There may exist a ceiling effect as picture naming was found to be significantly improved from a within patient median of 9/10 pre-ETV to a within patient median of 10/10 post-ETV, and a within patient change of +1 (0, +1), p = 0.0162. The significance was largely drawn from two patients who had +7 and +9 point improvements, all other patients had improvements of 0 or +1. Seven of ten patients achieved max scores, thus limiting the assessment of potential improved gained. Semantic fluency was found to not significantly changed, p = 0.3571. However, four patients had worse post-ETV semantic fluency scores. Three of these four patients achieved max scores in picture naming and the other scoring 9/10 both pre- and post-ETV. The patient with clinically worsened language index score achieved max score on picture name and -9 words on semantic fluency.

3.3.4 Retrospective prediction of global cognition improvement based on memory scores In iNPH, the RBANS was used pre- and post-external lumbar drain to predict cognitive improvement after shunting (Nakatsu et al., 2016). They found that when both baseline immediate and delayed memory index scores were below 80, they could retrospectively predict 13/19 patients who improved post-shunting. In our study, the difference between immediate and delayed memory index scores was within three points of each other in all patients who showed clinically significant improvement in the RBANS TS.

Remembering that the delayed memory index scores was based off recall and not recognition, with the knowledge that immediate memory and recall delayed memory are symptoms of subcortical dementia, it fits with the expected symptomatology of chronic obstructive hydrocephalus and these patients would therefore expect to improve. However, the three patients with worse delayed memory scores when compared to immediate memory present with an amnestic pattern of dementia. So, these patients do not fit with the subcortical dementia symptoms of chronic obstructive hydrocephalus and unlikely to improve in their cognitive symptoms post-ETV. The last group represent those with delayed memory scores greater than their immediate memory scores. This does not mean that they remembered more after a delay than immediately after presenting the stimuli, it is the relative difference in capacity. This is an uncommon symptom presentation that can be seen in those with high test anxiety or where questions of effort may exist. This dementia paradigm suggests that pre-ETV determination of those with frontosubcortical dementia symptomatology and chronic obstructive hydrocephalus may predict who will have global cognitive improvements post-ETV.

3.3.5 MoCA vs RBANS

In this cohort, the MoCA TS was not statistically or clinically changed. Whereas, the RBANS TS was both clinically and statistically improved. Individual analysis shows the MoCA TS was clinically worsened in 30% (3/10) and clinically improved in 20% (2/10). The RBANS TS had 50% (5/10) of patients clinically improved with no worsening. This finding highlights the benefits of extended neurocognitive testing. Individual analysis found no trend in MoCA subdomains leading to the clinical worsening or improvement.

Chapter 4: Conclusions

4.1 Summary of Thesis Objectives

This thesis aimed to determine the efficacy of ETV to improve cognition in chronic obstructive hydrocephalus. The literature to date consists of retrospective, underpowered, post-operative, global cognitive assessments (Al-Jumaily et al., 2012; Burtscher et al., 2003; Hader et al., 2014; Lacy et al., 2009; Takahashi, 2006; Azab et al., 2016 for review.) The published evidence for cognitive improvement post-ETV is weak and conflicted. So, we designed two studies accounting for those previous limitations with the following objectives:

O1. To determine the sufficiency of primary ETV to safely improve global cognition in adults with chronic obstructive hydrocephalus.

O2. To determine which cognitive domains ETV produces clinically meaningful changes in adults with chronic obstructive hydrocephalus.

4.2 **Objective 1: Conclusions and Limitations**

Primary ETV can safely improve symptoms of cognitive dysfunction in adults with chronic obstructive hydrocephalus. ETV resulted in significant and clinically meaningful long-term improvement in global cognition in some adults with chronic obstructive hydrocephalus. Deterioration of global cognition was rare following ETV. ETV is an effective treatment for patients with chronic obstructive hydrocephalus and may be superior to shunt procedures given the potentially lower failure and complication rates of ETVs. Future evaluations of the AHCRN registry patients undergoing ETV should further strengthen these observations.

A limitation of this study comes from incomplete registry data relevant both to patients who were too impaired to undergo assessment either at baseline (36 patients), 5-months post-ETV (13 patients), and due to a small number of patients with only either a pre- or post-ETV MoCA. An additional limitation is the modest population size. However, the results were obtained using standardized methodology for cognitive assessment with within patient analysis, thus maximizing the power of the sample size.

4.3 **Objective 2: Conclusions and Limitations**

Patients with chronic obstructive hydrocephalus treated with ETV have a clinically and statistically significant improvement in global cognition, attention and delayed memory cognitive domains at 4-months follow-up. There was no worsening in global cognition, attention, delayed memory, or immediate memory indexes. Although there were no statistically worsened cognitive domains, 50% of patients had a clinical worsening in visuospatial index scores. A potential predictor of post-ETV global cognitive improvement was discovered. The likelihood of RBANS TS clinical improvement was 100% when the difference between immediate and delayed memory indexes was three or less. Further validation of this predictor is required. Future studies should expand on the neurocognitive testing to include anatomical correlates in an effort to elucidate the mechanism of cognitive decline seen in chronic obstructive hydrocephalus.

The largest limitation of this study is the exploratory nature of the statistics and small sample size. Larger studies are required to validate the results. Special attention should be given to minimize the chance of selection bias. Longer follow-up is also essential to determine how these

cognitive domains change over time in order to determine the sufficiency of ETV to provide long-term symptom relief. The digit span subtest has previously been noted to often have a ceiling effect when tested on "normal" populations, leading to the potential for it to be a less robust subtest for measuring the cognitive construct (Costello & Obsborne, 2005). Despite this potential ceiling effect, our population was sufficiently poor pre-ETV that none of the patients achieved maximum scores, but two patients did achieve maximum scores by improving by +3 and +4 points at the four-month follow-up. Even though in our analysis statistical and clinical significance was reached, it should be noted that this is a possible limitation of this subtest. The picture naming subtest may also have a ceiling effect as 70% of patient achieve max scores post-ETV. Additionally, MCIDs have not been validated in this population and may not be have high accuracy to clinical change in the visuospatial domain.

References

- Ades-Aron B, Yeager S, Miskin N, Fieremans E, George A, Golomb J. Diffusional Kurtosis along the Corticospinal Tract in Adult Normal Pressure Hydrocephalus. *AJNR Am J Neuroradiol.* 2018;39(12):2218-2223. doi:10.3174/ajnr.A5845
- Ajilore O, Lamar M, Medina J, Watari K, Elderkin-Thompson V, Kumar A. Disassociation of verbal learning and hippocampal volume in type 2 diabetes and major depression. *Int J Geriatr Psychiatry*. 2015;30(4):393-399. doi:10.1002/gps.4149
- Al-Jumaily MJB, Hayhurst C, Jenkinson MD, Murphey P, Buxton N, Mallucci C (2012) Long term neuropsychological outcome and management of decompensated longstanding overt ventriculomegaly in adults. Br J Neurosurg 26(5):717–721. https://doi.org/10.3109/02688697.2012.673647
- Anandh B, Madhusudan Reddy KR, Mohanty A, Umamaheswara Rao GS, Chandramouli BA. Intraoperative bradycardia and postoperative hyperkalemia in patients undergoing endoscopic third ventriculostomy. Minim Invasive Neurosurg. 2002;45:154–157.
- Andrén, K., Wikkelsø, C., Tisell, M., & Hellström, P. (2014). Natural course of idiopathic normal pressure hydrocephalus. *Journal of Neurology, Neurosurgery & Psychiatry*, 85(7), 806-810.
- Antoniades M, Schoeler T, Radua J, et al. Verbal learning and hippocampal dysfunction in schizophrenia: A meta-analysis. *Neurosci Biobehav Rev.* 2018;86:166-175. doi:10.1016/j.neubiorev.2017.12.001
- Aslaksen PM, Bystad MK, Ørbo MC, Vangberg TR. The relation of hippocampal subfield volumes to verbal episodic memory measured by the California Verbal Learning Test II in healthy adults. *Behav Brain Res.* 2018;351:131-137. doi:10.1016/j.bbr.2018.06.008

- Assaf Y, Ben-Sira L, Constantini S, Chang LC, Beni-Adani L. Diffusion tensor imaging in hydrocephalus: initial experience. *AJNR Am J Neuroradiol*. 2006;27(8):1717-1724.
- Avagliano, L., Massa, V., George, T. M., Qureshy, S., Bulfamante, G. P., & Finnell, R. H.
 (2019). Overview on neural tube defects: From development to physical characteristics. Birth Defects Research, 111(19), 1455-1467. doi:10.1002/bdr2.1380
- Ayaz Hossain M, Frampton AE, Choo M, et al. The role of a vascular access surgeon in ventriculo-venous shunts in difficult hydrocephalus. *J Vasc Access*. 2010;11(2):150-154. doi:10.1177/112972981001100212
- Azab, W. A., Mijalcic, R. M., Nakhi, S. B., & Mohammad, M. H. (2016). Ventricular volume and neurocognitive outcome after endoscopic third ventriculostomy: is shunting a better option? A review. *Child's Nervous System*, 32(5), 775–780. <u>http://doi.org/10.1007/s00381-</u> <u>016-3032-3</u>
- Baddelely, A. D., & Hitch, G. (1974). Working memory. In G. H. Bower (Ed.), *The psychology of learning and motivation* (Vol. 8, pp. 47–90). San Diego: Academic Press.
- Bähner F, Demanuele C, Schweiger J, et al. Hippocampal-dorsolateral prefrontal coupling as a species-conserved cognitive mechanism: a human translational imaging study. *Neuropsychopharmacology*. 2015;40(7):1674-1681. doi:10.1038/npp.2015.13
- Barnett A, Marty-Dugas J, McAndrews MP. Advantages of sentence-level fMRI language tasks in presurgical language mapping for temporal lobe epilepsy. *Epilepsy Behav*. 2014;32:114-120. doi:10.1016/j.yebeh.2014.01.010
- Baroncini M, Kuchcinski G, Le Thuc V, et al. Is endoscopic third ventriculostomy safe and efficient in the treatment of obstructive chronic hydrocephalus in adults? A prospective

clinical and MRI study. *Acta Neurochir (Wien)*. 2019;161(7):1353-1360. doi:10.1007/s00701-019-03932-2.

- Baykan N, Isbir O, Gercek A, Dagcnar A, Ozek MM : Ten years of experience with pediatric neuroendoscopic third ventriculostomy: features and perioperative complications of 210 cases. J Neurosurg Anesthesiol 17 : 33-37, 2005
- Benabarre, A., Ibanez, J., Boget, T., Obiols, J., Martinez-Aran, A., & Vieta, E. (2001).
 Neuropsychological and psychiatric complications in endoscopic third ventriculostomy: a clinical case report. *Journal of Neurology, Neurosurgery & Psychiatry*, 71(2), 268-271.
- Benedict RH, DeLuca J, Phillips G, et al. Validity of the Symbol Digit Modalities Test as a cognition performance outcome measure for multiple sclerosis. *Mult Scler*. 2017;23(5):721-733. doi:10.1177/1352458517690821.
- Bernard R, Vallee F, Mateo J, et al. Uncontrollable high-frequency tachypnea: a rare and nearly fatal complication of endoscopic third ventriculostomy: case report and literature review. *Minim Invasive Neurosurg*. 2010;53(5-6):270-272. doi:10.1055/s-0030-1269874
- Bi Y, Wei T, Wu C, Han Z, Jiang T, Caramazza A. The role of the left anterior temporal lobe in language processing revisited: Evidence from an individual with ATL resection. *Cortex*. 2011;47(5):575-587. doi:10.1016/j.cortex.2009.12.002
- Biesbroek JM, van Zandvoort MJ, Kuijf HJ, et al. The anatomy of visuospatial construction revealed by lesion-symptom mapping. *Neuropsychologia*. 2014;62:68-76. doi:10.1016/j.neuropsychologia.2014.07.013
- Binder JR, Desai RH, Graves WW, Conant LL. Where is the semantic system? A critical review and meta-analysis of 120 functional neuroimaging studies. *Cereb Cortex*. 2009;19(12):2767-2796. doi:10.1093/cercor/bhp055

- Binder JR, Gross WL, Allendorfer JB, et al. Mapping anterior temporal lobe language areas with fMRI: a multicenter normative study. *Neuroimage*. 2011;54(2):1465-1475. doi:10.1016/j.neuroimage.2010.09.048
- Bonanni R, Carlesimo GA, Caltagirone C. Amnesia following endoscopic third ventriculostomy: a single case study. *Eur Neurol*. 2004;51(2):118-120. doi:10.1159/000076793
- Bor-Seng-Shu, E., Figueiredo, E. G., Fonoff, E. T., Fujimoto, Y., Panerai, R. B., & Teixeira, M.
 J. (2013). Decompressive craniectomy and head injury: brain morphometry, ICP, cerebral hemodynamics, cerebral microvascular reactivity, and neurochemistry. *Neurosurgical Review*, *36*(3), 361-370.
- Borgbjerg BM, Gjerris F, Albeck MJ, Hauerberg J, Børgesen SV. A comparison between ventriculo-peritoneal and ventriculo-atrial cerebrospinal fluid shunts in relation to rate of revision and durability. *Acta Neurochir (Wien)*. 1998;140(5):459-465. doi:10.1007/s007010050125
- Borghei A, Cothran T, Brahimaj B, Sani S. Role of massa intermedia in human neurocognitive processing. *Brain Struct Funct*. 2020;225(3):985-993. doi:10.1007/s00429-020-02050-5
- Bouras T, Sgouros S : Complications of endoscopic third ventriculostomy. J Neurosurg Pediatr 7 : 643-649, 2011
- Bouras T, Sgouros S : Complications of endoscopic third ventriculostomy. World Neurosurg 79(Suppl):S22.e9-e12, 2013
- Bräutigam, K., Vakis, A., & Tsitsipanis, C. (2019). Pathogenesis of idiopathic normal pressure hydrocephalus: a review of knowledge. Journal of Clinical Neuroscience, 61, 10-13.
- Brown MW, Banks PJ. In search of a recognition memory engram. *Neurosci Biobehav Rev*. 2015;50:12-28. doi:10.1016/j.neubiorev.2014.09.016

- Bruijnen CJ, Dijkstra BA, Walvoort SJ, et al. Psychometric properties of the Montreal Cognitive Assessment (MoCA) in healthy participants aged 18–70. Int J Psychiatry Clin Pract. 2020;1-8. doi: 10.1080/13651501.2020.1746348.
- Burtscher J, Bartha L, Twerdy K, et al. Effect of endoscopic third ventriculostomy on neuropsychological outcome in late onset idiopathic aqueduct stenosis: a prospective study. J. Neurol. Neurosurg. Psychiatry. 2003;74(2):222-225. doi:10.1136/jnnp.74.2.222.
- Bushnell, M. C., Duncan, G. H., Hofbauer, R. K., Ha, B., Chen, J. I., & Carrier, B. (1999). Pain perception: is there a role for primary somatosensory cortex?. *Proceedings of the National Academy of Sciences*, 96(14), 7705-7709.
- Camina, E., & Güell, F. (2017). The Neuroanatomical, Neurophysiological and Psychological Basis of Memory: Current Models and Their Origins. Frontiers in pharmacology, 8, 438. https://doi.org/10.3389/fphar.2017.00438
- Cao B, Passos IC, Mwangi B, et al. Hippocampal volume and verbal memory performance in late-stage bipolar disorder. *J Psychiatr Res*. 2016;73:102-107.
 doi:10.1016/j.jpsychires.2015.12.012
- Carlozzi NE, Horner MD, Yang C, Tilley BC. Factor analysis of the Repeatable Battery for the Assessment of Neuropsychological Status. *Appl Neuropsychol*. 2008;15(4):274-279. doi:10.1080/09084280802325124
- Chistyakov, A. V., Hafner, H., Sinai, A., Kaplan, B., & Zaaroor, M. (2012). Motor cortex disinhibition in normal-pressure hydrocephalus. *Journal of Neurosurgery*, 116(2), 453-459.
- Clarke E. & O'Malley (1996). The Human Brain and Spinal Cord: A Historical Study Illustrated by Writings from Antiquity to the Twentieth Century. Norman Publishing.

Clarke E. & O'Malley C.D. (1996) The human brain and spinal cord: A historical study illustrated by writings from antiquity to the twentieth century. Norman Publishing

Cohen, R. A. (1993). The neuropsychology of attention. New York: Plenum Press.

- Collmann H, Sörensen N, Krauss J. Hydrocephalus in craniosynostosis: a review. *Childs Nerv Syst.* 2005;21(10):902-912. doi:10.1007/s00381-004-1116-y
- Costello, Anna B. and Osborne, Jason (2005). Best practices in exploratory factor analysis: four recommendations for getting the most from your analysis. Practical Assessment, Research, and Evaluation: Vol. 10, Article 7. DOI: https://doi.org/10.7275/jyj1-4868
- Craven C, Asif H, Farrukh A, Somavilla F, Toma AK, Watkins L. Case series of ventriculopleural shunts in adults: a single-center experience. *J Neurosurg*. 2017;126(6):2010-2016. doi:10.3171/2016.4.JNS16641
- Damle, N. R., Ikuta, T., John, M., Peters, B. D., DeRosse, P., Malhotra, A. K., & Szeszko, P. R. (2017). Relationship among interthalamic adhesion size, thalamic anatomy and neuropsychological functions in healthy volunteers. *Brain structure & function*, 222(5), 2183– 2192. https://doi.org/10.1007/s00429-016-1334-6
- de Kunder SL, Ter Laak-Poort MP, Nicolai J, Vles JS, Cornips EM : Fever after intraventricular neuroendoscopic procedures in children. **Childs Nerv Syst 32** : 1049-1055, 2016
- de Kunder SL, Ter Laak-Poort MP, Nicolai J, Vles JS, Cornips EM. Fever after intraventricular neuroendoscopic procedures in children. Childs Nerv Syst. 2016;32:1049–1055.
- Del Bigio, M. R. (1998). Epidemiology and direct economic impact of hydrocephalus: A community based study. Canadian Journal of Neurological Sciences, 25(2), 123–126. https://doi.org/10.1017/S0317167100033722

- Demetriades AK, Haq IZ, Jarosz J, McCormick D, Bassi S. The ventriculocholecystic shunt: two case reports and a review of the literature. *Br J Neurosurg*. 2013;27(4):505-508. doi:10.3109/02688697.2013.771135
- Dennis EL, Disner SG, Fani N, et al. Altered white matter microstructural organization in posttraumatic stress disorder across 3047 adults: results from the PGC-ENIGMA PTSD consortium [published online ahead of print, 2019 Dec 19]. *Mol Psychiatry*.
 2019;10.1038/s41380-019-0631-x. doi:10.1038/s41380-019-0631-x
- Drew, T., & Vogel, E. K. (2009). Working memory: Capacity limitations. In L. R. Squire (Ed.), Encyclopedia of neuroscience (pp. 523-531). Oxford, England: Academic Press.
- Duff K, Hobson VL, Beglinger LJ, O'Bryant SE. Diagnostic accuracy of the RBANS in mild cognitive impairment: limitations on assessing milder impairments. *Arch Clin Neuropsychol*. 2010;25(5):429-441. doi:10.1093/arclin/acq045
- Duff K, Langbehn DR, Schoenberg MR, et al. Examining the repeatable battery for the assessment of neuropsychological status: factor analytic studies in an elderly sample. *Am J Geriatr Psychiatry*. 2006;14(11):976-979. doi:10.1097/01.JGP.0000229690.70011.cd
- Duff K, Schoenberg MR, Patton D, et al. Regression-based formulas for predicting change in RBANS subtests with older adults. *Arch Clin Neuropsychol*. 2005;20(3):281-290. doi:10.1016/j.acn.2004.07.007
- Duff K, Schoenberg MR, Patton D, Mold J, Scott JG, Adams RL. Predicting change with the RBANS in a community dwelling elderly sample. *J Int Neuropsychol Soc*. 2004;10(6):828-834. doi:10.1017/s1355617704106048
- Dunn, L. T. (2002). Raised intracranial pressure. *Journal of Neurology, Neurosurgery & Psychiatry*, 73(suppl 1), i23-i27.

- Dusick JR, McArthur DL, Bergsneider M. Success and complication rates of endoscopic third ventriculostomy for adult hydrocephalus: a series of 108 patients. *Surg Neurol.* 2008;69(1):5-15. doi:10.1016/j.surneu.2007.08.024.
- Emmert N, Schwarz L, Vander Wal J, Gfeller J. RBANS factor structure in older adults with suspected cognitive impairment: Evidence for a 5-factor structure. *Appl Neuropsychol Adult*. 2018;25(1):38-50. doi:10.1080/23279095.2016.1238827
- Fakhoury, M., & Domínguez López, S. (2014). The role of habenula in motivation and reward. Advances in Neuroscience, 2014.
- Fattahian, R., Bagheri, S. R., & Sadeghi, M. (2018). Development of Posttraumatic
 Hydrocephalus Requiring Ventriculoperitoneal Shunt After Decompressive Craniectomy for
 Traumatic Brain Injury: a Systematic Review and Meta-analysis of Retrospective
 Studies. *Medical archives (Sarajevo, Bosnia and Herzegovina)*, 72(3), 214–219.
 https://doi.org/10.5455/medarh.2018.72.214-219
- Flores, Leandro Pretto. (2002). Occipital lobe morphological anatomy: anatomical and surgical aspects. Arquivos de Neuro-Psiquiatria, 60(3A), 566-571. https://dx.doi.org/10.1590/S0004-282X2002000400010
- Folstein MF, Folstein SE, McHugh PR. "Mini-mental state". A practical method for grading the cognitive state of patients for the clinician. *J Psychiatr Res.* 1975;12(3):189-198. doi:10.1016/0022-3956(75)90026-6
- Friederici AD. White-matter pathways for speech and language processing. *Handb Clin Neurol*. 2015;129:177-186. doi:10.1016/B978-0-444-62630-1.00010-X

- Fujii M, Maesawa S, Ishiai S, Iwami K, Futamura M, Saito K. Neural Basis of Language: An Overview of An Evolving Model. *Neurol Med Chir (Tokyo)*. 2016;56(7):379-386. doi:10.2176/nmc.ra.2016-0014
- Garcia C, Leahy B, Corradi K, Forchetti C. Component structure of the Repeatable Battery for the Assessment of Neuropsychological Status in dementia. *Arch Clin Neuropsychol.* 2008;23(1):63-72. doi:10.1016/j.acn.2007.08.008
- Gerrish, A. C., Thomas, A. G., & Dineen, R. A. (2014, October). Brain white matter tracts:functional anatomy and clinical relevance. In Seminars in Ultrasound, CT and MRI (Vol. 35, No. 5, pp. 432-444). WB Saunders.
- Giussani C, Guida L, Trezza A, Sganzerla EP. Effectiveness of Intraventricular Endoscopic Lamina Terminalis Fenestration in Comparison with Standard ETV: Systematic Review of Literature. *World Neurosurg*. 2017;103:257-264. doi:10.1016/j.wneu.2017.03.143
- Gmeiner, M., Wagner, H., Schlögl, C., van Ouwerkerk, W. J., Senker, W., Sardi, G., ... &
 Gruber, A. (2019). Adult Outcome in Shunted Pediatric Hydrocephalus: Long-Term
 Functional, Social, and Neurocognitive Results. *World Neurosurgery*, *132*, e314-e323.
- Goldman-Rakic, P. S. (1987). Circuitry of the prefrontal cortex and the regulation of behavior by representational memory. In F. Plum (Ed.), Handbook of physiology: Section 1. The nervous system (pp. 373-417). Bethesda, MD: American Physiological Society.
- Golomb J, de Leon MJ, George AE et al. Hippocampal atrophy correlates with severe cognitive impairment in elderly patients with suspected normal pressure hydrocephalus *J Neurol Neurosurg Psych.* 1994;57:590-593.
- Gomes-de-Souza L, Costa-Ferreira W, Oliveira LA, Benini R, Crestani CC. Cannabinoid receptor type 1 in the bed nucleus of the stria terminalis modulates cardiovascular responses

to stress via local N-methyl-D-aspartate receptor/neuronal nitric oxide synthase/soluble guanylate cyclase/protein kinase G signaling. J Psychopharmacol. 2020;34(4):429-440. doi:10.1177/0269881119897556

- Goode TD, Acca GM, Maren S. Threat imminence dictates the role of the bed nucleus of the stria terminalis in contextual fear. *Neurobiol Learn Mem.* 2020;167:107116. doi:10.1016/j.nlm.2019.107116
- Greenwood J Jr. Paraphysial cysts of the third ventricle; with report of eight cases. *J Neurosurg*. 1949;6(2):153-159. doi:10.3171/jns.1949.6.2.0153
- Hader WJ, Brooks BL, Partlo L, Hamilton M (2014) Neuropsychological outcome after endoscopic third ventriculostomy. Can J Neurol Sci 41(6):729–734. https://doi.org/10.1017/cjn.2014.108
- Hallaert, G. G., Vanhauwaert, D. J., Logghe, K., Van den Broecke, C., Baert, E., Van Roost, D.,
 & Caemaert, J. (2012). Endoscopic coagulation of choroid plexus hyperplasia: Case report.
 Journal of Neurosurgery: Pediatrics, 9(2), 169-177.
- Halperin JJ, Kurlan R, Schwalb JM, Cusimano MD, Gronseth G, Gloss D. Practice guideline:
 Idiopathic normal pressure hydrocephalus: Response to shunting and predictors of response:
 Report of the Guideline Development, Dissemination, and Implementation Subcommittee of
 the American Academy of Neurology [published correction appears in Neurology. 2016 Feb
 23;86(8):793]. *Neurology*. 2015;85(23):2063-2071. doi:10.1212/WNL.00000000002193
- Hamada, H., Hayashi, N., Kurimoto, M., Takaiwa, A., Kurosaki, K., & Endo, S. (2009).
 Neuropsychological changes after endoscopic third ventriculostomy for long-standing overt ventriculomegaly in adults. *Neurologia medico-chirurgica*, 49(8), 362-364.

- Handler MH, Abbott R, Lee M. A near-fatal complication of endoscopic third ventriculostomy: case report. *Neurosurgery*. 1994;35(3):525-528. doi:10.1227/00006123-199409000-00025
- Hattingen E, Jurcoane A, Melber J, et al. Diffusion tensor imaging in patients with adult chronic idiopathic hydrocephalus. *Neurosurgery*. 2010;66(5):917-924.
 doi:10.1227/01.NEU.0000367801.35654.EC
- Hattori T, Yuasa T, Aoki S, et al. Altered microstructure in corticospinal tract in idiopathic normal pressure hydrocephalus: comparison with Alzheimer disease and Parkinson disease with dementia. *AJNR Am J Neuroradiol*. 2011;32(9):1681-1687. doi:10.3174/ajnr.A2570
- Hayreh SS. The sheath of the optic nerve. *Ophthalmologica*. 1984;189(1-2):54-63. doi:10.1159/000309386
- Hiraoka, K., Yamasaki, H., Takagi, M., Saito, M., Nishio, Y., Iizuka, O., ... & Mori, E. (2011). Is the midbrain involved in the manifestation of gait disturbance in idiopathic normal-pressure hydrocephalus?. *Journal of Neurology*, 258(5), 820-825.
- Holden HM, Milano NJ, Horner MD. Five-factor structure of the RBANS is supported in an Alzheimer's disease sample: Implications for validation of neuropsychological assessment instruments. *Appl Neuropsychol Adult*. 2020;27(3):232-242.
 doi:10.1080/23279095.2018.1529671
- Hong, J., Barrena, B. G., Lollis, S. S., & Bauer, D. F. (2016). Surgical management of arrested hydrocephalus: case report, literature review, and 18-month follow-up. *Clinical neurology and neurosurgery*, 151, 79-85.
- Ibrahim AK. Urinary bladder stone complicating ventriculovesical shunt. *Sultan Qaboos Univ Med J.* 2014;14(1):e142-e144. doi:10.12816/0003351

- Iddon JL, Pickard JD, Cross JJ et al. Specific patterns of cognitive impairment in patients with idiopathic normal pressure hydrocephalus and Alzheimer's disease: a pilot study. *J Neurol Neurosurg Psych*.1999;67(6):723-732.
- Irie R, Tsuruta K, Hori M, et al. Neurite orientation dispersion and density imaging for evaluation of corticospinal tract in idiopathic normal pressure hydrocephalus. *Jpn J Radiol.* 2017;35(1):25-30. doi:10.1007/s11604-016-0594-7
- Irwin, R. S., & Rippe, J. M. (Eds.). (2008). Irwin and Rippe's intensive care medicine. Lippincott Williams & Wilkins. Chapter 177.
- Isaacs AM, Bezchlibnyk YB, Yong H, et al. Endoscopic third ventriculostomy for treatment of adult hydrocephalus: long-term follow-up of 163 patients. *Neurosurg Focus*. 2016;41(3):E3. doi:10.3171/2016.6.FOCUS16193.
- Isaacs, A. M., Riva-Cambrin, J., Yavin, D., Hockley, A., Pringsheim, T. M., Jette, N., Lethebe,
 B. C., Lowerison, M., Dronyk, J., & Hamilton, M. G. (2018). Age-specific global
 epidemiology of hydrocephalus: Systematic review, metanalysis and global birth
 surveillance. *PloS one*, *13*(10), e0204926. https://doi.org/10.1371/journal.pone.0204926
- Iwańczuk, W., & Guźniczak, P. (2015). Neurophysiological foundations of sleep, arousal, awareness and consciousness phenomena. Part 1. *Anaesthesiology intensive therapy*, 47(2), 162.
- Jang SH, Ho Kim S. Diffusion tensor imaging following shunt in a patient with hydrocephalus. *J Neuroimaging*. 2011;21(1):69-72. doi:10.1111/j.1552-6569.2009.00394.x
- Jones, R. F. C., Stening, W. A., & Brydon, M. (1990). Endoscopic third ventriculostomy. Neurosurgery, 26(1), 86-92.

- Jung TY, Chong S, Kim IY, et al. Prevention of Complications in Endoscopic Third Ventriculostomy. J Korean Neurosurg Soc. 2017;60(3):282-288. doi:10.3340/jkns.2017.0101.014
- Jurcoane A, Keil F, Szelenyi A, Pfeilschifter W, Singer OC, Hattingen E. Directional diffusion of corticospinal tract supports therapy decisions in idiopathic normal-pressure hydrocephalus. *Neuroradiology*. 2014;56(1):5-13. doi:10.1007/s00234-013-1289-8
- Kamiya K, Hori M, Irie R, et al. Diffusion imaging of reversible and irreversible microstructural changes within the corticospinal tract in idiopathic normal pressure hydrocephalus. *Neuroimage Clin.* 2017;14:663-671. Published 2017 Mar 11. doi:10.1016/j.nicl.2017.03.003
- Karunanayaka PR, Holland SK, Schmithorst VJ, et al. Age-related connectivity changes in fMRI data from children listening to stories. *Neuroimage*. 2007;34(1):349-360. doi:10.1016/j.neuroimage.2006.08.028
- Kawsar KA, Haque MR, Chowdhury FH : Avoidance and management of perioperative complications of endoscopic third ventriculostomy: the Dhaka experience. J Neurosurg 123 : 1414-1419, 2015
- Kazui H, Miyajima M, Mori E, et al., Lumboperitoneal shunt surgery for idiopathic normal pressure hydrocephalus (SINPHONI-2): an open-label randomised trial. *Lancet Neurol*. 2015;14(6):585–594. doi:10.1016/S1474-4422(15)00046-0
- Kent PL. Working Memory: A Selective Review. *Appl Neuropsychol Child*. 2016;5(3):163-172. doi:10.1080/21622965.2016.1167491
- King LC, Bailie JM, Kinney DI, Nitch SR. Is the repeatable battery for the assessment of neuropsychological status factor structure appropriate for inpatient psychiatry? An

exploratory and higher-order analysis. *Arch Clin Neuropsychol*. 2012;27(7):756-765. doi:10.1093/arclin/acs062

- Kinoshita Y, Tominaga A, Saitoh T, Usui S, Takayasu T, Arita K, et al. Postoperative fever specific to neuroendoscopic procedures. Neurosurg Rev: 2013. ;doi:10.1007/s10143-013-0505-7
- Klinge P, Hellström P, Tans J, Wikkelsø C; European iNPH Multicentre Study Group. One-year outcome in the European multicentre study on iNPH. *Acta Neurol Scand*. 2012;126(3):145– 153. doi:10.1111/j.1600-0404.2012.01676.x
- Kondziella D, Eyjolfsson EM, Saether O, Sonnewald U, Risa O. Gray matter metabolism in acute and chronic hydrocephalus. Neuroscience. 2009;159(2):570-577. doi:10.1016/j.neuroscience.2009.01.008
- Kremen, W. S., Seidman, L. J., Faraone, S. V., Pepple, J. R., & Tsuang, M. T. (1992).
 Attention/information-processing factors in psychotic disorders: Replication and extension of recent neuropsychological findings. *Journal of Nervous and Mental Disease, 180*(2), 89–93. https://doi.org/10.1097/00005053-199202000-00004
- Krishnan K, Rossetti H, Hynan LS, et al. Changes in Montreal Cognitive Assessment Scores Over Time. *Assessment*. 2017;24(6):772-777. doi:10.1177/1073191116654217.
- Krishnan SS, Nigam P, Manuel A, Vasudevan MC. Lumbo-peritoneal shunt as treatment modality for visual symptoms in idiopathic intracranial hypertension: A single institution experience [published online ahead of print, 2020 May 29]. *Neurocirugia (Astur)*.
 2020;S1130-1473(20)30040-3. doi:10.1016/j.neucir.2020.03.001

- Kulkarni AV, Riva-Cambrin J, Holubkov R, Browd SR, Cochrane DD, Drake JM, et al : Endoscopic third ventriculostomy in children: Prospective, multicenter results from the hydrocephalus clinical research network. **J Neurosurg Pediatr 18** : 423-429, 2016
- Kulkarni AV, Schiff SJ, Mbabazi-Kabachelor E, et al. Endoscopic Treatment versus Shunting for Infant Hydrocephalus in Uganda. *N Engl J Med.* 2017;377(25):2456-2464. doi:10.1056/NEJMoa1707568
- Labidi M, Lavoie P, Lapointe G, et al. Predicting success of endoscopic third ventriculostomy: validation of the ETV Success Score in a mixed population of adult and pediatric patients. *J Neurosurg*. 2015;123(6):1447-1455. doi:10.3171/2014.12.JNS141240.
- Lacy M, Oliveira M, Austria E, Frim MD (2009) Neurocognitive outcome after endoscopic third ventriculocisterostomy in patients with obstructive hydrocephalus. J Int Neuropsychol Soc 15(3):394–398. https://doi.org/10.1017/S1355617709090560
- Larsson A, Stephensen H, Wikkelsø C. Adult patients with "asymptomatic" and "compensated" hydrocephalus benefit from surgery. *Acta Neurol Scand*. 1999;99(2):81-90. doi:10.1111/j.1600-0404.1999.tb00662.x
- Laurence KM: Neurological and intellectual sequelae of hydrocephalus. Arch Neurol 20:73-81, 1969.
- Lavrador, J. P., Ferreira, V., Lourenço, M., Alexandre, I., Rocha, M., Oliveira, E., ... & Neto, L.
 (2019). White-matter commissures: a clinically focused anatomical review. *Surgical and Radiologic Anatomy*, 1-12.
- Lee, C. Y., Chen, L. K., Lo, Y. K., Liang, C. K., Chou, M. Y., Lo, C. C., ... & Lin, Y. T. (2011). Urinary incontinence: an under-recognized risk factor for falls among elderly dementia patients. *Neurourology and Urodynamics*, 30(7), 1286-1290.

Lee, P. H., Yong, S. W., Ahn, Y. H., & Huh, K. (2005). Correlation of midbrain diameter and gait disturbance in patients with idiopathic normal pressure hydrocephalus. *Journal of Neurology*, 252(8), 958-963.

Lezak, M. D., Howieson, D. B., Bigler, E. D., & Tranel, D. (2012). Neuropsychological assessment (5th ed.). New York: Oxford University Press.

- Liu, J. T., & Su, P. H. (2020). The efficacy and limitation of lumboperitoneal shunt in normal pressure hydrocephalus. *Clinical Neurology and Neurosurgery*, *193*, 105748.
- Locatelli M, Draghi R, Di Cristofori A, et al. Third Ventriculostomy in Late-onset Idiopathic Aqueductal Stenosis Treatment: A Focus on Clinical Presentation and Radiological Diagnosis. *Neurol Med Chir (Tokyo)*. 2014;54(12):1014-1021. doi:10.2176/nmc.oa.2013-0367.
- Jugé, L., Pong, A. C., Bongers, A., Sinkus, R., Bilston, L. E., & Cheng, S. (2016). Changes in Rat Brain Tissue Microstructure and Stiffness during the Development of Experimental Obstructive Hydrocephalus. *PloS one*, *11*(2), e0148652. https://doi.org/10.1371/journal.pone.0148652
- Luyten L, Hendrickx S, Raymaekers S, Gabriëls L, Nuttin B. Electrical stimulation in the bed nucleus of the stria terminalis alleviates severe obsessive-compulsive disorder. *Mol Psychiatry*. 2016;21(9):1272-1280. doi:10.1038/mp.2015.124
- Maeda, M., Takachi, T., Nakai, M., Krieger, A. J., & Sapru, H. N. (1994). Medullary cardiovascular center and acute brain swelling. *In Brain Edema IX* (pp. 168-170). Springer, Vienna.

- Maller, J.J., Welton, T., Middione, M. *et al.* Revealing the Hippocampal Connectome through Super-Resolution 1150-Direction Diffusion MRI. *Sci Rep* 9, 2418 (2019).
 https://doi.org/10.1038/s41598-018-37905-9
- Mao J, Zhu Q, Ma Y, Lan Q, Cheng Y, Liu G. Fenestration of Lamina Terminalis During Anterior Circulation Aneurysm Clipping on Occurrence of Shunt-Dependent Hydrocephalus After Aneurysmal Subarachnoid Hemorrhage: Meta-Analysis. World Neurosurg. 2019;129:e1-e5. doi:10.1016/j.wneu.2019.01.270
- Marmarou A, Bergsneider M, Relkin N, Klinge P, Black PM. Development of guidelines for idiopathic normal-pressure hydrocephalus: introduction. *Neurosurgery*. 2005;57(3 Suppl):S1v. doi:10.1227/01.neu.0000168188.25559.0e
- Maule WF. Nausea and Vomiting. In: Walker HK, Hall WD, Hurst JW, editors. *Clinical Methods: The History, Physical, and Laboratory Examinations*. 3rd edition. Boston:
 Butterworths; 1990. Chapter 84.
- Mayo CD, Mazerolle EL, Ritchie L, Fisk JD, Gawryluk JR; Alzheimer's Disease Neuroimaging Initiative. Longitudinal changes in microstructural white matter metrics in Alzheimer's disease. *Neuroimage Clin.* 2016;13:330-338. Published 2016 Dec 16. doi:10.1016/j.nicl.2016.12.012
- McGrew, K. S. (2009). CHC theory and the human cognitive abilities project: Standing on the shoulders of giants of psychometric intelligence research. Intelligence, 37, 1-10.
- Mervis, C. B., Robinson, B. F., & Pani, J. R. (1999). Visuospatial construction. *American journal of human genetics*, 65(5), 1222–1229. https://doi.org/10.1086/302633
- Mesulam, M. M. (1981). A cortical network for directed attention and unilateral neglect. *Annals* of Neurology, 10, 309–325.

- Middlebrooks, E.H., Yagmurlu, K., Szaflarski, J.P. *et al.* A contemporary framework of language processing in the human brain in the context of preoperative and intraoperative language mapping. *Neuroradiology* **59**, 69–87 (2017). https://doi.org/10.1007/s00234-016-1772-0
- Miller, A. D., & Leslie, R. A. (1994). The area postrema and vomiting. *Frontiers in Neuroendocrinology*, 15(4), 301-320.
- Mirsky, A. F. (1996). Disorders of attention: A neuropsycho- logical perspective. In G. R. Lyon & N. A. Krasnegor (Eds.), *Attention, memory, and executive function* (pp. 71–96). Baltimore: Paul H. Brookes.
- Mirsky, A.F., Anthony, B.J., Duncan, C.C. *et al.* Analysis of the elements of attention: A neuropsychological approach. *Neuropsychol Rev* 2, 109–145 (1991). https://doi.org/10.1007/BF01109051
- Mocco, J., Tomey, M. I., Komotar, R. J., Mack, W. J., Frucht, S. J., Goodman, R. R., & McKhann, G. M. (2006). Ventriculoperitoneal shunting of idiopathic normal pressure hydrocephalus increases midbrain size: a potential mechanism for gait improvement. *Neurosurgery*, 59(4), 847-851.
- Monro A. Observations on structure and functions of the nervous system. Edinburgh: Creech and Johnson, 1783.
- Mori E, Ishikawa M, Kato T, et al. Guidelines for management of idiopathic normal pressure hydrocephalus: second edition. *Neurol Med Chir (Tokyo)*. 2012;52(11):775-809. doi:10.2176/nmc.52.775
- Nakatsu D, Fukuhara T, Chaytor NS, Phatak VS, Avellino AM. Repeatable Battery for the Assessment of Neuropsychological Status (RBANS) as a Cognitive Evaluation Tool for

Patients with Normal Pressure Hydrocephalus. *Neurol Med Chir (Tokyo)*. 2016;56(2):51-61. doi:10.2176/nmc.oa.2015-0027

- Nasreddine ZS, Phillips NA, Bédirian V, et al. The Montreal Cognitive Assessment, MoCA: A Brief Screening Tool For Mild Cognitive Impairment. J. Am. Geriatr. Soc. 2005;53(4):695-699. doi:10.1111/j.1532-5415.2005.53221.x.
- Nichols DE. N,N-dimethyltryptamine and the pineal gland: Separating fact from myth. J Psychopharmacol. 2018;32(1):30-36. doi:10.1177/0269881117736919
- Nilakantan AS, Mesulam MM, Weintraub S, Karp EL, VanHaerents S, Voss JL. Networktargeted stimulation engages neurobehavioral hallmarks of age-related memory decline. *Neurology*. 2019;92(20):e2349-e2354. doi:10.1212/WNL.000000000007502
- Nulsen FE, & Spitz EB: Treatment of hydrocephalus by direct shunt from ventricle to jugular vein. *Surg Forum* 399–403, 1951
- O'Connell ME, Gould B, Ursenbach J, Enright J, Morgan DG. Reliable change and minimum clinically important difference (MCID) of the Repeatable Battery for the Assessment of Neuropsychology Status (RBANS) in a heterogeneous dementia sample: Support for reliable change methods but not the MCID. *Appl Neuropsychol Adult*. 2019;26(3):268-274. doi:10.1080/23279095.2017.1413575
- Ogino A, Kazui H, Miyoshi N, et al. Cognitive impairment in patients with idiopathic normal pressure hydrocephalus. *Dement Geriatr Cogn Disord*. 2006;21(2):113-119. doi:10.1159/000090510
- Osuka S, Matsushita A, Yamamoto T, et al. Evaluation of ventriculomegaly using diffusion tensor imaging: correlations with chronic hydrocephalus and atrophy. *J Neurosurg*. 2010;112(4):832-839. doi:10.3171/2009.7.JNS09550

- Patwardhan, R. V., & Nanda, A. (2005). Implanted Ventricular Shunts in the United States: The Billion-dollar-a-year Cost of Hydrocephalus Treatment. Neurosurgery, 56(1), 139–145. http://doi.org/10.1227/01.NEU.0000146206.40375.41
- Peterson KA, Mole TB, Keong NCH, et al. Structural correlates of cognitive impairment in normal pressure hydrocephalus. *Acta Neurol Scand*. 2019;139(3):305-312. doi:10.1111/ane.13052
- Pham, A. C., Fan, C., & Owler, B. K. (2013). Treating pediatric hydrocephalus in Australia: a 3year hospital-based cost analysis and comparison with other studies. *Journal of neurosurgery*. *Pediatrics*, 11(4), 398–401. <u>https://doi.org/10.3171/2013.1.PEDS12233</u>
- Phillips R, Qi G, Collinson SL, et al. The Minimum Clinically Important Difference in the Repeatable Battery for the Assessment of Neuropsychological Status. *Clin Neuropsychol*. 2015;29(7):905-923. doi:10.1080/13854046.2015.1107137
- Pindrik J, Riva-Cambrin J, Kulkarni AV, et al. Surgical resource utilization after initial treatment of infant hydrocephalus: comparing ETV, early experience of ETV with choroid plexus cauterization, and shunt insertion in the Hydrocephalus Clinical Research Network [published online ahead of print, 2020 Jun 19]. *J Neurosurg Pediatr*. 2020;1-9. doi:10.3171/2020.4.PEDS19632
- Pong, A. C., Jugé, L., Bilston, L. E., & Cheng, S. (2017). Development of acute hydrocephalus does not change brain tissue mechanical properties in adult rats, but in juvenile rats. *PloS* one, 12(8), e0182808. <u>https://doi.org/10.1371/journal.pone.0182808</u>
- Posner, M. I., & Petersen, S. E. (1990). The attention system of the human brain. *Annual Review* of Neuroscience, 13, 25–42.

- Randolph C, Tierney MC, Mohr E, Chase TN. The Repeatable Battery for the Assessment of Neuropsychological Status (RBANS): preliminary clinical validity. J Clin Exp Neuropsychol. 1998;20(3):310-319. doi:10.1076/jcen.20.3.310.823
- Reddy GK, Bollam P, Caldito G. Long-term outcomes of ventriculoperitoneal shunt surgery in patients with hydrocephalus. *World Neurosurg*. 2014;81(2):404-410. doi:10.1016/j.wneu.2013.01.096.
- Rigamonti, D. (Ed.). (2014). Adult hydrocephalus. Cambridge University Press. Chapter 8.
- Saether, K., Hilaire, G., & Monteau, R. (1987). Dorsal and ventral respiratory groups of neurons in the medulla of the rat. Brain research, 419(1-2), 87-96.
- Satyarthee G. D. (2018). Postdecompressive Craniectomy Surgery, Ventriculomegaly, or Hydrocephalus Development: Imaging, Prevention, and Management. *Journal of neurosciences in rural practice*, 9(2), 177–179. https://doi.org/10.4103/jnrp.jnrp_547_17
- Scheel, M., Diekhoff, T., Sprung, C., & Hoffmann, K. T. (2012). Diffusion tensor imaging in hydrocephalus—findings before and after shunt surgery. *Acta neurochirurgica*, 154(9), 1699-1706.
- Schiller F. The cerebral ventricles. From soul to sink. *Arch Neurol*. 1997;54(9):1158-1162. doi:10.1001/archneur.1997.00550210086018
- Schirmer, C. M. (2007). Mechanisms of visual loss in papilledema. *Neurosurgical Focus*, 23(5), E5-E5.
- Schmitt AL, Livingston RB, Smernoff EN, Reese EM, Hafer DG, Harris JB. Factor analysis of the Repeatable Battery for the Assessment of Neuropsychological Status (RBANS) in a large sample of patients suspected of dementia. *Appl Neuropsychol*. 2010;17(1):8-17. doi:10.1080/09084280903297719

- Schroeder HW, Niendorf WR, Gaab MR : Complications of endoscopic third ventriculostomy. J Neurosurg 96 : 1032-1040, 2002
- Senova S, Fomenko A, Gondard E, Lozano AM. Anatomy and function of the fornix in the context of its potential as a therapeutic target. *J Neurol Neurosurg Psychiatry*. 2020;91(5):547-559. doi:10.1136/jnnp-2019-322375
- Sherman, D., Fuller, P. M., Marcus, J., Yu, J., Zhang, P., Chamberlin, N. L., ... & Lu, J. (2015). Anatomical location of the mesencephalic locomotor region and its possible role in locomotion, posture, cataplexy, and parkinsonism. *Frontiers in Neurology*, 6, 140.
- Silva PS, Cerejo A, Polónia P, Pereira J, Vaz R. Trans-lamina terminalis approach for third ventricle and suprasellar tumours. Clin Neurol Neurosurg. 2013;115(9):1745-1752. doi:10.1016/j.clineuro.2013.04.010
- Smith, A. *Symbol digit modalities test: manual*. Los Angeles, CA: Western Psychological Services; 1982:21-22
- Soleman J, Guzman R. Neurocognitive Complications after Ventricular Neuroendoscopy: A Systematic Review. Behav Neurol. 2020;2536319. doi:10.1155/2020/2536319
- Song, G., Yu, Y., & Poon, C. S. (2006). Cytoarchitecture of pneumotaxic integration of respiratory and nonrespiratory information in the rat. *Journal of Neuroscience*, 26(1), 300-310.
- Spennato P, Mirone G, Nastro A, et al. Hydrocephalus in Dandy-Walker malformation. *Childs Nerv Syst.* 2011;27(10):1665-1681. doi:10.1007/s00381-011-1544-4
- Standring, S. (2008) Gray's Anatomy: The Anatomical Basis of Clinical Practice. 40th Edition, Churchill Livingstone Elsevier, London.

- Strauss, E., Sherman, E. M. S., & Spreen, O. (2006). A compendium of neuropsychological tests: Administration, norms, and commentary (3rd ed.). Oxford University Press.
- Strauss, M. E., Thompson, P., Adams, N. L., Redline, S., & Burant, C. (2000). Evaluation of a model of attention with confirmatory factor analysis. *Neuropsychology*, 14(2), 201– 208. <u>https://doi.org/10.1037/0894-4105.14.2.201</u>

Sundström, N., Malm, J., Laurell, K., Lundin, F., Kahlon, B., Cesarini, K. G., Leijon, G., & Wikkelsö, C. (2017). Incidence and outcome of surgery for adult hydrocephalus patients in Sweden. *British journal of neurosurgery*, *31*(1), 21–27. https://doi.org/10.1080/02688697.2016.1229749

- Switka A, Narkiewicz O, Dziewiatkowski J, Moryś J. The shape of the inferior horn of the lateral ventricle in relation to collateral and occipitotemporal sulci. Folia Morphol (Warsz). 1999;58(2):69-80.
- Takahashi Y (2006) Long-term outcome and neurologic develop- ment after endoscopic third ventriculostomy versus shunting during infancy. Childs Nerv Syst 22(12):1591–1602. https://doi.org/10.1007/s00381-006-0190-8
- Tanaka, M., Osada, T., Ogawa, A., Kamagata, K., Aoki, S., & Konishi, S. (2020). Dissociable Networks of the Lateral/Medial Mammillary Body in the Human Brain. *Frontiers in human neuroscience*, 14, 228. https://doi.org/10.3389/fnhum.2020.00228
- Tate MC, Herbet G, Moritz-Gasser S, Tate JE, Duffau H. Probabilistic map of critical functional regions of the human cerebral cortex: Broca's area revisited. *Brain*. 2014;137(Pt 10):2773-2782. doi:10.1093/brain/awu168

- Taveira KV, Carraro KT, Catalão CH, Lopes Lda S. Morphological and morphometric analysis of the hippocampus in Wistar rats with experimental hydrocephalus. Pediatr Neurosurg. 2012;48(3):163-167. doi:10.1159/000345959
- Tekin S, Cummings JL. Frontal-subcortical neuronal circuits and clinical neuropsychiatry: an update. *J Psychosom Res.* 2002;53(2):647-654. doi:10.1016/s0022-3999(02)00428-2
- Teo C. Third Ventriculostomy in the Treatment of Hydrocephalus: Experience with More than 120 Cases. In: *Minimally InvasiTechniques for Neurosurgery*. Berlin, Heidelberg: Springer Berlin Heidelberg; 1998:73-76. doi:10.1007/978-3-642-58731-3_11.
- Torrence, N. D., John, S. E., Gavett, B. E., & O'Bryant, S. E. (2016). An empirical comparison of competing factor structures for the Repeatable Battery for the Assessment of Neuropsychological Status: A Project FRONTIER Study. *Archives of Clinical Neuropsychology*, 31(1), 88-96.
- Torres-Corzo J.G., Rangel-Castilla L., Nakaji P. (2016). Neuroendoscopic Surgery. Thieme Verlagsgruppe, Stuttgart, New York, Delhi, Rio.
- Tubbs, R. S., Oakes, P., Maran, I. S., Salib, C., & Loukas, M. (2014). The foramen of monro: A review of its anatomy, history, pathology, and surgery. Child's Nervous System, 30(10), 1645-1649. doi:10.1007/s00381-014-2512-6
- Tudor KI, Tudor M, McCleery J, Car J. Endoscopic third ventriculostomy (ETV) for idiopathic normal pressure hydrocephalus (iNPH). *Cochrane Database Syst Rev.* 2015;(7):CD010033.
 Published 2015 Jul 29. doi:10.1002/14651858.CD010033.pub2
- Tuite, G. F., Chong, W. K., Evanson, J., Narita, A., Taylor, D., Harkness, W. F., ... & Hayward,R. D. (1996). The effectiveness of papilledema as an indicator of raised intracranial pressurein children with craniosynostosis. *Neurosurgery*, 38(2), 272-278.
- van Aalst J, Beuls EAM, Luijckx GJ. Neuropsychological and psychiatric complications in endoscopic third ventriculostomy. *Journal of Neurology, Neurosurgery & Psychiatry* 2002;73:460.
- Vogt EM, Prichett GD, Hoelzle JB. Invariant two-component structure of the Repeatable Battery for the Assessment of Neuropsychological Status (RBANS). *Appl Neuropsychol Adult*. 2017;24(1):50-64. doi:10.1080/23279095.2015.1088852
- Vries, J. K. (1978). An endoscopic technique for third ventriculostomy. Surgical Neurology, 9(3), 165.
- Wang JX, Rogers LM, Gross EZ, et al. Targeted enhancement of cortical-hippocampal brain networks and associative memory. *Science*. 2014;345(6200):1054-1057. doi:10.1126/science.1252900
- Waqar M, Ellenbogen JR, Stovell MG, et al. Long-Term Outcomes of Endoscopic Third Ventriculostomy in Adults. *WNEU*. 2016;94:386-393. doi:10.1016/j.wneu.2016.07.028.
- Wilde MC. The validity of the repeatable battery of neuropsychological status in acute stroke. *Clin Neuropsychol*. 2006;20(4):702-715. doi:10.1080/13854040500246901
- Williams MA, Nagel SJ, Luciano MG, et al. The clinical spectrum of hydrocephalus in adults: report of the first 517 patients of the Adult Hydrocephalus Clinical Research Network registry. *J Neurosurg*. 2019;34(aop):1-12. doi:10.3171/2019.2.JNS183538.
- Wu EM, El Ahmadieh TY, Kafka B, et al. Ventriculoperitoneal Shunt Outcomes of Normal Pressure Hydrocephalus: A Case Series of 116 Patients. *Cureus*. 2019;11(3):e4170. Published 2019 Mar 4. doi:10.7759/cureus.4170

- Yagmurlu K, Middlebrooks EH, Tanriover N, Rhoton AL Jr. Fiber tracts of the dorsal language stream in the human brain. *J Neurosurg*. 2016;124(5):1396-1405. doi:10.3171/2015.5.JNS15455
- Yamada, S. M., Tomita, Y., Murakami, H., Nakane, M., Yamada, S., Murakami, M., ... & Matsuno, A. (2018). Headache in patients with chronic subdural hematoma: analysis in 1080 patients. *Neurosurgical Review*, 41(2), 549-556.
- Yang, C., Garrett-Mayer, E., Schneider, J. S., Gollomp, S. M., & Tilley, B. C. (2009).
 Repeatable battery for assessment of neuropsychological status in early Parkinson's disease. *Movement disorders: official journal of the Movement Disorder Society*, 24(10), 1453-1460.
- Yuan W, Holland SK, Shimony JS, et al. Abnormal structural connectivity in the brain networks of children with hydrocephalus. *Neuroimage Clin*. 2015;8:483-492. Published 2015 Apr 29. doi:10.1016/j.nicl.2015.04.015
- Zhu, X. L., Gao, R., Wong, G. K. C., Wong, H. T., Ng, R. Y. T., Yu, Y., ... & Poon, W. S. (2013). Single burr hole rigid endoscopic third ventriculostomy and endoscopic tumor biopsy: What is the safe displacement range for the foramen of Monro?. *Asian Journal of Surgery*, *36*(2), 74-82.
- Zillmer E.A., Spiers M.V., Culbertson W.C. (2008) Principles of Neuropsychology, Second Edition. Belmont, California : Thomson-Wadsworth

Appendices Appendix A Tables

Table 1. MoCA Total Score and Symbol Digit Modalities Test (SDMT)											
	Follow-up	Sample Size (n)	Pre-ETV Median [Q1-Q3]	Post-ETV Median [Q1-Q3]	Within Patient Change [Q1-Q3]	P-value	Clinical Improvement n (%)	Clinical Worsening n (%)			
MoCA Total Score	5 Months	38	24 [23-27]	26 [24-28]	+1 [0-2]	< 0.001	15 (40%)	2 (5%)			
	14 Months	15	23 [22- 27]	26 [25-28]	+2 [1-3]	0.007	8 (53%)	1 (7%)			
SDMT	5 Months	32	39 [29-48]	45 [35-50]	+3 [-1-6]	0.084	14 (44%)	9 (28%)			
	14 Months	14	32 [28-43]	38 [30-46]	+3 [1-8]	0.138	7 (50%)	3 (21%)			

Table 2. Neurocognitive assessments: MoCA and RBANS Domains											
	Pre-ETV Median [Q1-Q3]	Post-ETV Median [Q1-Q3]	Within Patient Change [Q1-Q3]	P-value	Clinical Improvement n (%)	Clinical Worsening n (%)					
MoCA Total Score	24 [21-27]	24 [18-25]	0.00 [-2.00, +1.00]	0.7204	2 (20%)	3 (30%)					
RBANS Total Score	72 [55-81]	77 [66-89]	+8.0 [+4.25, +15.50]	0.0125	5 (50%)	0					
RBANS Total Scale Percentile	3.5 [0.33- 11]	6.5 [1.25- 23.5]	+3.45 [+0.30, +18.00]	0.0284	N/A	N/A					
RBANS: Immediate Memory Index Score	76.5 [66- 84.5]	81.5 [65- 101.2]	$^{+2.50}_{[0.00, +14.25]}$	0.1322	4 (40%)	0					
RBANS: Visuospatial Constructional Index Score	98.0 [63.75- 107.75]	84.9 [73.5- 98.0]	+5.5 [-3.0, +9.0]	0.3850	2 (20%)	5 (50%)					
RBANS: Language Index Score	88.0 [57.00- 95.25]	85.5 [76.0,105.0]	+5.0 [-6.50, +16.5]	0.2399	5 (50%)	1 (10%)					
RBANS: Attention Index Score	70.0 [56.75- 80.25]	71.5 [64.0, 98.5]	+10 [+3.25, +19]	0.0058	7 (70%)	0					
RBANS: Delayed memory Index Score	75.5 [55.0, 83.0]	92.5 [76.25, 98.75]	+11.50 [+4.50, +20.25]	0.0143	5 (50%)	0					



Figure 1. Flow diagram of patient inclusions, exclusions and follow-up. MoCA: Montreal Cognitive Assessment. ETV: Endoscopic Third Ventriculostomy.



Figure 2. MoCA Total Score for 5-Months Follow-up (n=38; left) and 14-Months Follow-up (n=15; right)

<u>5-Months Follow-up</u>: Median pre-ETV MoCA score of 24/30 (orange bar) improved to 26/30 post ETV (green bar). Median within patient change was +1 point (black bar; p < 0.001). Dotted lines show a clinically significant change of $\ge \pm 2$ points (Krishnan et al., 2017).

<u>14-Months Follow-up</u>: Median pre-ETV MoCA of 23/30 (orange bar) improved to 26/30 post ETV (green bar). Median within patient change was +2 points (black bar; p = 0.007), which is also clinically significant.



Figure 3. Flow diagram of patient inclusions, exclusions and follow-up in Chapter 3 study.



Figure 4. MoCA Total Score, not significantly changed at 4-Months Follow-up. Each column contains data one time point. N = 9 for all time points. Dotted line (A & B) shows lowest normal score of 26/30. Red bins represent clinically worsened scores. Green bin represents the clinically improved score. Clinically significant change is $\geq \pm 2$ points (Krishnan et al., 2017).



Figure 5. Statistically significant changes in primary outcomes: RBANS Total Score, Total Score Percentiles or index scores (attention and delayed memory). Each column contains data one time point. Each row displays histograms for one score. Change scores that meet the MCID are highlighted in orange. N = 10 for all scores.



Figure 6. Correlation between Pre-ETV Total Score Percentile and Change in Total Scale Percentile. The blue line is the linear model $(x \sim y)$ regression line. The grey shading is the standard error based on the linear model.



Difference in PreDM-IM to Predict Post-ETV RBANS TS Improvement

Figure 7: The difference between pre-ETV delayed memory and immediate memory index scores accurately predict clinically significant improvement in post-ETV RBANS TS (MCID \geq 8). Vertical lines divide the differences between delay memory and immediate memory index scores into three groups: amnestic (left), subcortical dementia (center), and uncommon combination (right). Horizontal line indicates MCID for RBANS TS \geq 8. The dots above the horizontal line and between the two vertical lines are the five patients that had improved RBANS TS post-ETV. They also had very similar delayed and immediate memory scores, with three patients having the same scores, one with a difference of one, and the third patient with a difference of three points. Those with larger differences to either side failed to improve in RBANS TS post-ETV. This may be indicative of a different pathology causing the global cognitive dysfunction, which has a greater effect on either delayed or immediate memory.